

Dated 21 May

2021 BW

Planning Agreement

Environmental Planning and Assessment Act 1979

Minister for Planning and Public Spaces (ABN 20 770 707 468) Hydro Aluminium Kurri Kurri Pty Ltd (ABN 55 093 266 221)

SWATWK HIDDA

Kuhlz

Sydney . Melbourne . Brisbane . Caims www.holdingredlich.com

S:10585026_2 BXW

Table of contents

1.	Defin	itions and interpretation	3
	1.1 1.2	Definitions Interpretation	
2.	Opera	ation and application of this deed	9
	2.1 2.2 2.3	Operation Planning agreement under the Act Application	10
3.	Appli	cation of sections 7.11, 7.12 and 7.24 of the Act	10
4.	Deve	lopment Contributions	10
	4.1 4.2	Developer to provide Development Contributions Acknowledgement	
5.	Intere	est	10
	5.1	Interest for late payment	10
6.	Enfor	cement	11
	6.1 6.2	Compulsory Acquisition Developer to provide Security	
7.	Regis	tration	11
	7.1 7.2 7.3 7.4	Registration of deed Evidence of registration Release and discharge of deed Developer's interest in Land	11 12
8.	Dispu	te Resolution	12
	8.1 8.2 8.3 8.4 8.5 8.6 8.7	Not commence Written notice of dispute Attempt to resolve Mediation Court proceedings Not use information No prejudice	12 12 13 13 13
9.	GST		13
	9.1 9.2 9.3 9.4 9.5 9.6 9.7 9.8	Definitions Intention of the parties Reimbursement Consideration GST exclusive Additional Amounts for GST Non monetary consideration Assumptions No merger	13 13 14 14 14 14
10.	Assig	nment and transfer	14
	10.1 10.2 10.3	Right to assign or novate Right to transfer Land Replacement Security	15

•

and an

11.	Capa	city	. 16
	11.1		
	11.2	Power of attorney	
12.	Repo	rting requirement	. 16
13.	Gene	ral Provisions	. 17
	13.1	Entire deed	. 17
	13.2	Variation	. 17
	13.3	Waiver	. 17
	13.4	Further assurances	
	13.5	Time for doing acts	
	13.6	Governing law and jurisdiction	
	13.7	Severance	
	13.8	Preservation of existing rights	
	13.9	No merger	
) Counterparts	
		. Relationship of parties Good faith	
		No fetter	
		Explanatory note	
		Expenses and stamp duty	
		5 Notices	
Sche			
Sche	dule 2	Address for Service	. 22
Sche	dule 3	Land	. 23
Sche	dule 4	Development Contributions	. 24
Sche	dule 5	Security terms	. 33
Sche	dule 6	Remediation Works	. 35
Sche	dule 7	Scope of Works	. 42
		Independent Engineer's Deed	
Exect	ution p	age	. 44
Anne	exure A	Containment Cell Plan	. 45
Anne	exure B	Containment Cell Land Plan	. 46
Anne	exure C	Access Road Concept Plan	. 47
Anne	exure D	Access Road Certificate of Compliance	. 48
Anne	exure E	Stage 1A Works Certificate of Compliance	. 49

This Planning Agreement is dated

21 May 2021 2020 Brett Whiteof

Parties:

Minister

Minister for Planning and Public Spaces (ABN 20 770 707 468) of Level 15, 52 Martin Place, Sydney NSW 2000

Developer

Hydro Aluminium Kurri Kurri Pty Ltd (ABN 55 093 266 221) of Hart Road, Loxford NSW 2326

Luhi

Introduction:

A The Developer owns the Land.

- **B** The Developer has lodged a Development Application for Development Consent to carry out the Development on the Land.
- **C** The Developer has offered to enter into this deed with the Minister to provide the Development Contributions in connection with the Development.

It is agreed:

1. Definitions and interpretation

1.1 Definitions

In this **deed**, unless the context clearly indicates otherwise:

Access Easement means an easement over the Access Road Land, if required, for the purpose of providing the Minister and the owner or occupier of the Containment Cell Land a right of access to and from the Containment Cell Land by any means and the nearest public road, being Hart Road, Loxford.

Access Road means the road to be constructed by the Developer providing access to the Containment Cell Land, within the potential area for the Access Road identified on the Access Road Concept Plan, with the final location to be determined by the Access Road Plan, in accordance with clause 3.4 of Schedule 4 of this deed.

Access Road Certificate of Compliance means a certificate from an appropriately qualified engineer in the form attached as Annexure D.

Access Road Concept Plan means the plan attached at Annexure C as amended from time to time in accordance with clause 3.4 of Schedule 4 of this deed.

Access Road Land means:

(a) during the period prior to the Minister's approval of the Access Road Plan in accordance with clause 3.4(g)(i) of Schedule 4 of this deed, that part of the Land identified on the Access Road Concept Plan as the potential area for the Access Road; and (b) during the period after the Minister's approval of the Access Road Plan in accordance with clause 3.4(g)(i) of Schedule 4 of this deed, that part of the Land identified on the Access Road Plan as the area for the Access Road.

Act means the Environmental Planning and Assessment Act 1979 (NSW).

Address for Service means the address of each party appearing in Schedule 2 of this deed or any new address notified by any party to all other parties as its new Address for Service.

Approval means any approval, authorisation, consent, licence, permit, exemption, certificate or any other approval required by law.

Authority means any Federal, State or local government or semi-governmental, statutory, judicial or public person, instrumentality or department.

Bank Guarantee means an irrevocable and unconditional undertaking:

- (a) by an Australian bank which is an eligible financial institution for the purposes of Treasury Circular NSW TC14/01 dated 24 January 2014 as amended, supplemented or substituted from time to time; and
- (b) on terms acceptable to the Minister, in the Minister's absolute discretion,

to pay the face value of that undertaking (being such amount as is required under this deed) on demand.

Base CPI means the CPI number for the quarter ending 31 March 2020.

Business Day means any day that is not a Saturday, Sunday, public holiday or bank holiday in Sydney, and concludes at 5pm on that day.

Claim includes a claim, demand, remedy, suit, injury, damage, loss, cost, liability, action, proceeding or right of action.

CLM Act means the Contaminated Land Management Act 1997 (NSW).

Commencement Date means the date this deed commences in accordance with clause 2.1 of this deed.

Completion of Remediation Works means the date on which a Site Audit Statement and Site Audit Report are issued in relation to the Remediation Works.

Consent Authority has the same meaning as in the Act.

Construction Certificate has the same meaning as in the Act.

Construction Remediation Works means so much of the Remediation Works as relates to:

- (a) the construction of the Containment Cell, including all design and temporary works, the construction of infrastructure associated with the Containment Cell (such as the installation of pipework for collection of groundwater, collection sumps for leachate and 'anchor' trenches) and the construction of the access road to the Containment Cell and the perimeter road, as described in SSD 6666; and
- (b) the Stage 1B Works.

Containment Cell means the containment cell and associated infrastructure proposed to be constructed by the Developer on the Land in accordance with any Remediation Consent and generally as shown on the plan at Annexure A.

Containment Cell Condition Report means the report prepared by the Developer in accordance with clause 3.3 of Schedule 4 of this deed.

Containment Cell Contribution means the:

- (a) management and maintenance of the Containment Cell up to and during the Management Period in accordance with the LTEMP; and
- (b) construction of the Access Road,

in accordance with clause 3 of Schedule 4 of this deed.

Containment Cell Insurance means the insurances taken out and maintained by the Developer in accordance with clause 3.7 of Schedule 4 of this deed.

Containment Cell Land means that part of the Land to be transferred to the Minister, including the land on which the Containment Cell is located, in accordance with the terms of this deed and as shown on the Containment Cell Land Plan.

Containment Cell Land Plan means the plan attached as Annexure B to this deed.

Containment Cell Land Subdivision Plan means any Plan of Subdivision which creates a separate lot for the Containment Cell Land prepared in accordance with clause 3.4 of Schedule 4 of this deed.

Containment Cell Land Transfer Date means the date on which the Developer is to transfer the Containment Cell Land to the Minister in accordance with Schedule 4 of this deed.

Contamination has the same meaning as in the CLM Act.

CoRD Holder Consent means the electronic document lodged through an ELNO that provides consent to the registration of instruments and plans.

Costs means any loss, cost, fee, charge, expense, Tax, rate, fine, penalty or debts including those in connection with advisors and any compensation payable to any person in accordance with the law.

CPI means the Sydney Consumer Price Index (All Groups) published by the Commonwealth Statistician, or if that index no longer exists, any similar index that the Minister specifies, in his or her sole discretion, for the purposes of this deed.

CPI Adjustment Date means 1 July 2021 and each anniversary of 1 July 2021 thereafter.

Current CPI means the CPI number for the quarter ending on 31 March in the relevant adjustment year.

Date for Completion means the date 4 years from the date the Remediation Consent is granted.

Department means the NSW Department of Planning, Industry and Environment.

Development means the development of the Land generally in accordance with SSD 6666, comprising remediation of the former Hydro Kurri Kurri Aluminium smelter site including excavation of onsite contaminated areas, excavation and treatment of capped waste stockpile (CWS) material, construction of a purpose built containment cell, placement of contaminated

materials in the containment cell, treatment of CWS leachate and the contaminated groundwater plume beneath the CWS and ongoing management of the containment cell in perpetuity.

Development Application has the same meaning as in the Act.

Development Consent has the same meaning as in the Act.

Development Contributions means the contributions to be provided by the Developer in accordance with Schedule 4 and Schedule 6 of this deed.

ELNO has the same meaning as in the Electronic Conveyancing National Law (NSW).

Explanatory Note means the note exhibited with a copy of this deed when this deed is made available for inspection by the public pursuant to the Act, as required by the Regulation.

Fit for Purpose means, in respect of any part of the Remediation Works, fit for the purposes stated in or to be reasonably inferred from this deed (and the documents referenced therein including but not limited to the Remediation Consent and the RAP).

General Register of Deeds means the land register maintained under the *Conveyancing Act 1919* (NSW) and so titled.

GST means any form of goods and services tax payable under the GST Legislation.

GST Legislation means the A New Tax System (Goods and Services Tax) Act 1999 (Cth).

Independent Engineer means an appropriately qualified engineer engaged by the Developer in accordance with clause 3 of Schedule 6 of this deed.

Independent Engineer's Deed means the deed between the Minister, the Developer and the Independent Engineer substantially in the form that appears at Schedule 8 of this deed (with any changes to be agreed, in writing, by the Minister, acting reasonably).

Land means the land described in Schedule 3 of this deed.

LTEMP means the Long Term Environmental Management Plan prepared by the Developer in accordance with the Remediation Consent.

Management Period means a period of not less than 5 years commencing on the Remediation Works Completion Date.

Mediation Program means the Mediation Program of the Law Society of New South Wales as published on its website and as varied from time to time.

Minister means the NSW Minister for Planning and Public Spaces and includes (where relevant) the Minister's nominee, the Secretary and the Secretary's nominee.

Monetary Contribution means the monetary contribution to be provided by the Developer in accordance with Schedule 4 of this deed for the purpose of ensuring environmental protection measures for the perpetual care of the Containment Cell and associated infrastructure on the Containment Cell Land.

NSW EPA means the NSW Environment Protection Authority.

Planning Application means:

(a) a Development Application; or

(b) any other application required under the Act.

Plan of Subdivision means a registered plan of subdivision within the meaning of section 195 of the *Conveyancing Act 1919* (NSW).

Public Road has the same meaning as in the Roads Act 1993 (NSW).

Private Road has the same meaning as in the Roads Act 1993 (NSW).

Proposed Dealing has the meaning given in clause 3.4 of Schedule 4 of this deed.

RAP means the Remedial Action Plan - Hydro Aluminium Smelter Kurri Kurri prepared by Ramboll dated 2 July 2018 as amended from time to time (with the Minister's agreement).

Real Property Act means the Real Property Act 1900 (NSW).

Register means the Torrens title register maintained under the Real Property Act.

Regulation means the Environmental Planning and Assessment Regulation 2000 (NSW).

Remediation has the meaning given to it in *State Environmental Planning Policy No. 55 – Remediation of Land* and **remediate** has a corresponding meaning.

Remediation Consent means any Development Consent granted for the Development.

Remediation Criteria means the remediation criteria set out in the RAP.

Remediation Validation Report has the same meaning as in the Remediation Consent.

Remediation Works means all works to be undertaken in accordance with the Remediation Consent, including associated design, and, for the avoidance of doubt, includes the road that will form part of the Containment Cell Land.

Remediation Works Completion Date means the date on which the Minister provides the Developer with a notice in accordance with clause 3.2(f) of Schedule 4 of this deed.

Remediation Works Construction Contribution means the carrying out of the Remediation Works in accordance with the Remediation Consent and this deed by the Date for Completion.

Roads Authority has the same meaning as in the Roads Act 1993 (NSW).

Secretary means the Secretary of the Department.

Security means the Bank Guarantees for the amounts and on the terms specified in Schedule 5 and Schedule 6 of this deed.

Scope of Works means the scope of work for design and construction of the Remediation Works at Schedule 7 of this deed.

Site Auditor has the same meaning as in the CLM Act.

Site Audit Statement means a site audit statement as that term is defined in the CLM Act and that:

(a) has been prepared in accordance with the Remediation Consent; and

(b) is addressed to the Minister.

Site Audit Report means a site audit report as that term is defined in the CLM Act and that:

- (a) has been prepared in accordance with the Remediation Consent; and
- (b) is addressed to the Minister.

SSD 6666 means the State Significant Development Application No. SSD 6666 made by the Developer for the Development.

Stage 1A Works means the following works and activities forming part of SP2 Part 1 as outlined in the Scope of Works:

- (a) set-up of contractor's temporary project facilities;
- (b) installation of temporary fencing at work areas;
- (c) installation of soil and water management infrastructure at the site of the Containment Cell;
- (d) stockpiling of excavated material at temporary stockpiles for later use; and
- (e) clearing and excavation of clay borrow pit area for the Containment Cell.

Stage 1A Works Certificate of Compliance means a certificate from the Validation Consultant in the form attached as Annexure E.

Stage 1B Works means the following works and activities forming part of SP2 Part 1 as outlined in the Scope of Works:

- (a) creation of main East-West Haul Road;
- (b) creation of permanent creek crossing (culverts); and
- (c) construction of 5 dams 1 leachate dam, 3 sediment basins at Containment Cell site and 1 leachate dam at the capped waste stockpile.

Stage means each of the sections indicated as part of "SP2" in the Scope of Works at Schedule 7 of this deed.

Stage Completion has the meaning set out in clause 6 of Schedule 6 of this deed.

Suitably Qualified Consultant means a consultant certified under either the Environment Institute of Australia and New Zealand's Certified Environmental Practitioner (Site Contamination) scheme (CEnvP(SC)) or the Soil Science Australia Certified Professional Soil Scientist Contaminated Site Assessment and Management (CPSS/ CSAM) scheme.

Tax means a tax, duty (including stamp duty and any other transaction duty), levy, impost, charge, fee (including a registration fee) together with all interest, penalties, fines and costs concerning them.

Validation Consultant has the same meaning as in the Remediation Consent.

1.2 Interpretation

In this deed unless the context clearly indicates otherwise:

(a) a reference to **this deed** or another document means this deed or that other document and any document which varies, supplements, replaces, assigns or novates this deed or that other document;

- Page 9
- (b) a reference to **legislation** or a **legislative provision** includes any statutory modification, or substitution of that legislation or legislative provision and any subordinate legislation issued under that legislation or legislative provision;
- (c) a reference to a **body** or **authority** which ceases to exist is a reference to either a body or authority that the parties agree to substitute for the named body or authority or, failing agreement, to a body or authority having substantially the same objects as the named body or authority;
- (d) a reference to the **introduction**, a **clause**, a **schedule** or an **annexure** is a reference to the introduction, a clause, a schedule or an annexure to or of this deed;
- (e) **clause headings, the introduction** and the **table of contents** are inserted for convenience only and do not form part of this deed;
- (f) the schedules and annexures form part of this deed;
- (g) a reference to a **person** includes a natural person, corporation, statutory corporation, partnership, the Crown or any other organisation or legal entity;
- (h) a reference to a **natural person** includes their personal representatives, successors and permitted assigns;
- (i) a reference to a corporation includes its successors and permitted assigns;
- (j) a reference to a right or obligation of a party is a reference to a right or obligation of that party under this deed;
- (k) an **obligation** or **warranty** on the part of 2 or more persons binds them jointly and severally and an obligation or warranty in favour of 2 or more persons benefits them jointly and severally;
- (I) a requirement to do any thing includes a requirement to cause that thing to be done and a requirement not to do any thing includes a requirement to prevent that thing being done;
- (m) including and includes are not words of limitation;
- (n) a word that is derived from a defined word has a corresponding meaning;
- (o) monetary amounts are expressed in Australian dollars;
- (p) the singular includes the plural and vice-versa;
- (q) words importing one gender include all other genders;
- (r) a reference to a thing includes each part of that thing; and
- (s) neither this deed nor any part of it is to be construed against a party on the basis that the party or its lawyers were responsible for its drafting.

2. Operation and application of this deed

2.1 Operation

This deed commences on the date that this deed is signed by all the parties.

2.2 Planning agreement under the Act

This deed constitutes a planning agreement within the meaning of section 7.4 of the Act and the parties agree on the matters set out in Schedule 1 of this deed.

2.3 Application

This deed applies to:

- (a) the Land; and
- (b) the Development.

3. Application of sections 7.11, 7.12 and 7.24 of the Act

The application of sections 7.11, 7.12 and 7.24 of the Act is excluded to the extent stated in Schedule 1 of this deed.

4. Development Contributions

4.1 Developer to provide Development Contributions

The Developer undertakes to provide, or procure the provision of, the Development Contributions to the Minister subject to, and in accordance with, the provisions of Schedule 4 and Schedule 6 of this deed.

4.2 Acknowledgement

The Developer acknowledges and agrees that, subject to section 7.3 of the Act, the Minister:

- (a) has no obligation to use or expend the Development Contributions for a particular purpose despite any provision of this deed to the contrary and has no obligation to repay the Development Contributions; and
- (b) in circumstances where the Development Contributions are transferred to any Authority, has not made any representation or warranty that the Development Contributions will or must be used for a particular purpose by that Authority.
- (c) The Developer acknowledges and agrees that no approval or acceptance of any plan or specification by the Minister as required by this deed relieves the Developer of responsibility to properly design and construct the Remediation Works.

5. Interest

5.1 Interest for late payment

- (a) If the Developer fails to pay a monetary amount due to the Minister on the due date for payment, the Developer must also pay to the Minister interest at a rate of 2% above the loan reference rate charged by the Commonwealth Bank of Australia from time to time.
- (b) Interest is payable on the daily balance of amounts due from the due date for payment of those amounts until all outstanding amounts (including interest on those amounts) have been paid to the Minister.

6. Enforcement

6.1 Compulsory Acquisition

In addition to clause 6.2 below, the parties agree that clause 3.8 of Schedule 4 is the provision of security for the transfer of the Containment Cell Land to the Minister.

6.2 Developer to provide Security

The Developer has agreed to provide security to the Minister for the performance of the Developer's obligations under this deed by providing the Security to the Minister in accordance with the terms and procedures set out in Schedule 5 and clause 4 of Schedule 6.

7. Registration

7.1 Registration of deed

- Within 20 Business Days of receiving a copy of this deed executed by the Minister, the Developer at its own expense is to take all practical steps and otherwise do anything to procure:
 - (i) the consent of each person, as required by the Registrar-General, who:
 - (A) has an estate or interest in the Land registered under the Real Property Act; or
 - (B) is seized or possessed of an estate or interest in the Land,

to the registration of this deed on the title to the Land and to the terms of this deed;

- (ii) the execution of any documents reasonably required to procure the registration of this deed on the title to the Land;
- (iii) the production of the relevant certificates of title or electronic lodgement of the relevant CoRD Holder Consents through an ELNO; and
- (iv) the lodgement of this deed in a registrable form at the NSW Land Registry Services for registration by the Registrar-General in the relevant folio of the Register for the Land, or in the General Register of Deeds if this deed relates to land not under the Real Property Act.
- (b) The Developer will take all practical steps and otherwise do anything reasonably required to procure the registration of this deed within three months of the date of this deed in the relevant folio of the Register for the Land, or in the General Register of Deeds if this deed relates to land not under the Real Property Act, including promptly responding to any requisitions made by the Registrar-General in respect of this deed and/or any ancillary documents.

7.2 Evidence of registration

- (a) The Developer must provide the Minister with evidence of the lodgement of this deed pursuant to clause 7.1(a)(iv) within 10 Business Days of such lodgement at the NSW Land Registry Services.
- (b) The Developer will provide the Minister with a copy of the relevant folio of the Register for the Land and a copy of the registered dealing containing this deed within 10 Business Days of receipt of notice of registration of this deed.

7.3 Release and discharge of deed

- (a) The Minister agrees to do all things reasonably required by the Developer to release and discharge this deed with respect to any part of the Land (other than the Containment Cell Land and the Access Road Land) upon:
 - (i) the Developer paying the Monetary Contribution to the Minister; and
 - (ii) Completion of the Remediation Works with respect to that part of the Land.
 - (b) The Minister agrees to do all things reasonably required by the Developer to release and discharge this deed from the Access Road Land upon the satisfaction of clause 7.3(b)(i) or (ii) as follows (and subject to the Completion of Remediation Works in relation to the Access Road (as relevant) and payment of the Monetary Contribution):
 - to the extent that the Access Road is to be a Public Road, the Minister has provided the Developer with a notice in accordance with clause 3.4(g) of Schedule 4 of this deed and the Access Road has been dedicated to the Roads Authority as a Public Road; and
 - (ii) to the extent that the Access Road is a Private Road, the Minister has provided the Developer with a notice under clause 3.4(g) of Schedule 4 of this deed and the Access Easement has been registered over the Access Road Land.

7.4 Developer's interest in Land

The Developer represents and warrants that it is:

- (a) the owner of the Land identified in Schedule 3 of this deed; and
- (b) legally and beneficially entitled to obtain all consents and approvals and to compel any person referred to in or contemplated by clause 7.1(a)(i) to assist, cooperate and to otherwise do all things necessary for the Developer to comply with its obligations under clause 7.

8. Dispute Resolution

8.1 Not commence

A party must not commence any court proceedings relating to a dispute unless it complies with this clause 8.

8.2 Written notice of dispute

A party claiming that a dispute has arisen under or in relation to this deed must give written notice to the other party specifying the nature of the dispute.

8.3 Attempt to resolve

On receipt of notice under clause 8.2, the parties must endeavour in good faith to resolve the dispute expeditiously using informal dispute resolution processes such as mediation, expert evaluation or other methods agreed by them.

8.4 Mediation

If the parties do not agree within 21 Business Days of receipt of notice under clause 8.2 (or any further period agreed in writing by them) as to:

- (a) the dispute resolution technique and procedures to be adopted;
- (b) the timetable for all steps in those procedures; or
- (c) the selection and compensation of the independent person required for such technique,

the parties must mediate the dispute in accordance with the Mediation Program. The parties must request the president of the Law Society of NSW or the president's nominee to select the mediator and determine the mediator's remuneration.

8.5 Court proceedings

If the dispute is not resolved within 60 Business Days after notice is given under clause 8.2 then any party which has complied with the provisions of this clause 8 may in writing terminate any dispute resolution process undertaken under this clause and may then commence court proceedings in relation to the dispute.

8.6 Not use information

The parties acknowledge the purpose of any exchange of information or documents or the making of any offer of settlement under this clause 8 is to attempt to settle the dispute. No party may use any information or documents obtained through any dispute resolution process undertaken under this clause 8 for any purpose other than in an attempt to settle the dispute.

8.7 No prejudice

This clause 8 does not prejudice the right of a party to institute court proceedings for urgent injunctive or declaratory relief in relation to any matter arising out of or relating to this deed.

9. GST

9.1 Definitions

Words used in this clause that are defined in the GST Legislation have the meaning given in that legislation.

9.2 Intention of the parties

The parties intend that:

- (a) Divisions 81 and 82 of the GST Legislation apply to the supplies made under and in respect of this deed; and
- (b) no additional amounts will be payable on account of GST and no tax invoices will be exchanged between the parties.

9.3 Reimbursement

Any payment or reimbursement required to be made under this deed that is calculated by reference to a cost, expense, or other amount paid or incurred must be limited to the total cost, expense or amount less the amount of any input tax credit to which any entity is entitled for the acquisition to which the cost, expense or amount relates.

9.4 Consideration GST exclusive

Unless otherwise expressly stated, all prices or other sums payable or consideration to be provided under this deed are exclusive of GST. Any consideration that is specified to be inclusive of GST must not be taken into account in calculating the GST payable in relation to a supply for the purposes of this clause 9.4.

9.5 Additional Amounts for GST

- (a) To the extent an amount of GST is payable on a supply made by a party (**Supplier**) under or in connection with this deed (the **GST Amount**), the recipient must pay to the Supplier the GST Amount.
- (b) However, where a GST Amount is payable by the Minister as recipient of the supply, the Developer must ensure that:
 - (i) the Developer makes payment of the GST Amount on behalf of the Minister, including any gross up that may be required; and
 - (ii) the Developer provides a tax invoice to the Minister.

9.6 Non monetary consideration

Clause 9.5 applies to non-monetary consideration.

9.7 Assumptions

The Developer acknowledges and agrees that in calculating any amounts payable under clause 9.5 the Developer must assume the Minister is not entitled to any input tax credit.

9.8 No merger

This clause does not merge on completion or termination of this deed.

10. Assignment and transfer

10.1 Right to assign or novate

- (a) Prior to a proposed assignment or novation of its rights or obligations under this deed, the party seeking to assign its rights or novate its obligations (**Assigning Party**) must seek the consent of the Minister and:
 - satisfy the Minister (acting reasonably) that the person to whom the Assigning Party's rights or obligations are to be assigned or novated (**Incoming Party**) has sufficient assets, resources and expertise required to perform the Assigning Party's obligations under this deed insofar as those obligations are to be novated to the Incoming Party;
 - (ii) procure the execution of an agreement by the Incoming Party with the Minister on terms satisfactory to the Minister (acting reasonably) under which the Incoming Party agrees to comply with the terms and conditions of this deed as though the Incoming Party were the Assigning Party; and
 - (iii) satisfy the Minister, acting reasonably, that it is not in material breach of its obligations under this deed.

(b) The Assigning Party must pay the Minister's reasonable legal costs and expenses incurred under this clause 10.1.

10.2 Right to transfer Land

- (a) Subject to clause 10.2(b), the Developer must not sell or transfer to another person (**Transferee**) the whole or any part of the:
 - the Land prior to the payment of the Monetary Contribution in accordance with clause 2 of Schedule 4 of this deed and Completion of Remediation Works with respect to the relevant part of the Land;
 - the Access Road Land prior to the date on which the matters set out in clause
 7.3(b)(i) or (ii) are satisfied; or
 - (iii) the Containment Cell Land before the Containment Cell Land Transfer Date, even if the Developer has paid the Monetary Contribution in accordance with clause 2 of Schedule 4 of this deed and following Completion of Remediation Works.
- (b) The Developer may only sell or transfer the whole or any part of the Land, the Access Road Land, or the Containment Cell Land (as applicable) to a Transferee prior to satisfying the relevant matters set out in clause 10.2(a)(i) to (iii) if prior to the proposed sale or transfer the Developer:
 - satisfies the Minister, acting reasonably, that the proposed Transferee has sufficient assets, resources and expertise required to perform any of the remaining obligations of the Developer under this deed or satisfies the Minister, acting reasonably, that the Developer will continue to be bound by the terms of this deed after the transfer has been effected;
 - (ii) procures the execution of an agreement by the Transferee with the Minister on terms satisfactory to the Minister, acting reasonably, under which the Transferee agrees to comply with the terms and conditions of this deed as though the Transferee were the Developer; and
 - (iii) satisfies the Minister, acting reasonably, that it is not in material breach of its obligations under this deed.
- (c) The Developer must pay the Minister's reasonable legal costs and expenses incurred under this clause 10.2.
- (d) The Minister agrees not to sell or transfer its interest in the Containment Cell Land or to novate or assign its interest in the Planning Agreement other than to another Minister, NSW government agency, State owned corporation or local council, without the Developer's consent which cannot be unreasonably withheld.

10.3 Replacement Security

Provided that:

- (a) the Developer has complied with clauses 10.1 and 10.2; and
- (b) the Transferee or Incoming Party (as the case may be) has provided the Minister with a replacement Security in accordance with the requirements of Schedule 5 and Schedule 6 of this deed,

the Minister will promptly return the Security to the Developer.

11. Capacity

11.1 General warranties

Each party warrants to each other party that:

- (a) this deed creates legal, valid and binding obligations, enforceable against the relevant party in accordance with its terms; and
- (b) unless otherwise stated, it has not entered into this deed in the capacity of trustee of any trust.

11.2 Power of attorney

If an attorney executes this deed on behalf of any party, the attorney declares that it has no notice of the revocation of that power of attorney.

12. Reporting requirement

- (a) By 1 September each year until the Containment Cell Land Transfer Date or as otherwise agreed with the Secretary, the Developer must deliver to the Secretary a report (in a format acceptable to the Secretary) for the period 1 July to 30 June of the preceding financial year which must include the following matters, as applicable:
 - (i) details of all Development Consents and Subdivision Certificates issued in relation to the Development;
 - a description of the status of the Development including a plan that identifies what parts of the Development have been completed, are under construction and are to be constructed;
 - a forecast in relation to the anticipated progression and completion of the Development;
 - (iv) when the Developer expects to lodge the next Planning Application;
 - following the Remediation Works Completion Date, an assessment of the performance of the Containment Cell including whether it is performing satisfactorily; and
 - (vi) during the Remediation Works for the Development, a statement from the Site Auditor appointed for the Development stating that the Site Auditor has been appropriately briefed on the status of the Remediation Works and is satisfied that the remediation goals as set out in the RAP remain achievable.
- (b) Upon the Secretary's request, the Developer must deliver to the Secretary all documents and other information which, in the reasonable opinion of the Secretary, are necessary for the Secretary to assess the status of the Development and the Developer's compliance with this deed.

13. General Provisions

13.1 Entire deed

This deed constitutes the entire agreement between the parties regarding the matters set out in it and supersedes any prior representations, understandings or arrangements made between the parties, whether orally or in writing.

13.2 Variation

This deed must not be varied except by a later written document executed by all parties.

13.3 Waiver

A right created by this deed cannot be waived except in writing signed by the party entitled to that right. Delay by a party in exercising a right does not constitute a waiver of that right, nor will a waiver (either wholly or in part) by a party of a right operate as a subsequent waiver of the same right or of any other right of that party.

13.4 Further assurances

Each party must promptly execute all documents and do everything necessary or desirable to give full effect to the arrangements contained in this deed.

13.5 Time for doing acts

- (a) If:
- (i) the time for doing any act or thing required to be done; or
- (ii) a notice period specified in this deed,

expires on a day other than a Business Day, the time for doing that act or thing or the expiration of that notice period is extended until the following Business Day.

(b) If any act or thing required to be done is done after 5 pm on the specified day, it is taken to have been done on the following Business Day.

13.6 Governing law and jurisdiction

- (a) The laws applicable in New South Wales govern this deed.
- (b) The parties submit to the non-exclusive jurisdiction of the courts of New South Wales and any courts competent to hear appeals from those courts.

13.7 Severance

If any clause or part of any clause is in any way unenforceable, invalid or illegal, it is to be read down so as to be enforceable, valid and legal. If this is not possible, the clause (or where possible, the offending part) is to be severed from this deed without affecting the enforceability, validity or legality of the remaining clauses (or parts of those clauses) which will continue in full force and effect.

13.8 Preservation of existing rights

The expiration or termination of this deed does not affect any right that has accrued to a party before the expiration or termination date.

13.9 No merger

Any right or obligation of any party that is expressed to operate or have effect on or after the completion, expiration or termination of this deed for any reason, does not merge on the occurrence of that event but remains in full force and effect.

13.10 Counterparts

This deed may be executed in any number of counterparts. All counterparts taken together constitute one instrument.

13.11 Relationship of parties

Unless otherwise stated:

- (a) nothing in this deed creates a joint venture, partnership, or the relationship of principal and agent, or employee and employer between the parties; and
- (b) no party has the authority to bind any other party by any representation, declaration or admission, or to make any contract or commitment on behalf of any other party or to pledge any other party's credit.

13.12 Good faith

Each party must act in good faith towards all other parties and use its best endeavours to comply with the spirit and intention of this deed.

13.13 No fetter

Nothing in this deed is to be construed as requiring the Minister to do anything that would cause the Minister to breach any of the Minister's obligations at law and without limitation, nothing in this deed shall be construed as limiting or fettering in any way the discretion of the Minister in exercising any of the Minister's statutory functions, powers, authorities or duties.

13.14 Explanatory note

The Explanatory Note must not be used to assist in construing this deed.

13.15 Expenses and stamp duty

- (a) The Developer must pay its own and the Minister's reasonable legal costs and disbursements in connection with the negotiation, preparation, execution and carrying into effect of this deed and the Independent Engineer's Deed.
- (b) The Developer must pay for all reasonable costs and expenses associated with the giving of public notice of this deed and the Explanatory Note in accordance with the Regulation.
- (c) The Developer must pay all Taxes assessed on or in respect of this deed and any instrument or transaction required or contemplated by or necessary to give effect to this deed (including stamp duty and registration fees, if applicable).
- (d) The Developer must provide the Minister with bank cheques, or an alternative method of payment if agreed with the Minister, in respect of the Minister's costs pursuant to clauses 13.15(a) and (b):
 - (i) where the Minister has provided the Developer with written notice of the sum of such costs prior to execution, on the date of execution of this deed; or

(ii) where the Minister has not provided the Developer with prior written notice of the sum of such costs prior to execution, within 30 Business Days of demand by the Minister for payment.

13.16 Notices

- (a) Any notice, demand, consent, approval, request or other communication (**Notice**) to be given under this deed must be in writing and must be given to the recipient at its Address for Service (including as applicable the Landowner and Developer's Legal Advisor) by being:
 - (i) hand delivered; or
 - (ii) sent by facsimile transmission; or
 - (iii) sent by prepaid ordinary mail within Australia; or
 - (iv) in the case of a Notice to be given by the Developer, Minister or Secretary, sent by email.
- (b) A Notice is given if:
 - (i) hand delivered, on the date of delivery but if delivery occurs after 5pm New South Wales time or a day that is not a Business Day, is taken to be given on the next Business Day;
 - (ii) sent by facsimile and the sending party's facsimile machine reports that the facsimile has been successfully transmitted;
 - (A) before 5pm on a Business Day, on that day;
 - (B) after 5pm on a Business Day, on the next Business Day after it is sent; or
 - (C) on a day that is not a Business Day, on the next Business Day after it is sent; or
 - (iii) sent by prepaid ordinary mail within Australia, on the date that is 2 Business Days after the date of posting; or
 - (iv) sent by email:
 - (A) before 5pm on a Business Day, on that Day;
 - (B) after 5pm on a Business Day, on the next Business Day after it is sent; or
 - (C) on a day that it is not a Business Day, on the next Business Day after it is sent,

and the sender does not receive a delivery failure notice.

Schedule 1

Table 1 - Requirements under section 7.4 of the Act (clause 2.2)

The parties acknowledge and agree that the table set out below provides for certain terms, conditions and procedures for the purpose of the deed complying with the Act.

Requirement under the Act	This deed		
Planning instrument and/or development application – (section 7.4(2))			
The Developer has:			
 (a) sought a change to an environmental planning instrument. 	(a) No		
(b) made, or proposes to make, a Development Application.	(b) Yes		
 (c) entered into an agreement with, or is otherwise associated with, a person, to whom paragraph (a) or (b) applies. 	(c) No		
Description of land to which this deed applies – (section 7.4(3)(a))	See Schedule 3		
Description of development to which this deed applies – (section 7.4(3)(b))	See definition of Development in clause 1.1		
Description of change to the environmental planning instrument to which this deed applies – (section 7.4(3)(b))	N/A		
The scope, timing and manner of delivery of contribution required by this deed – (section 7.4(3)(c))	See Schedule 4 and Schedule 6		
Applicability of sections 7.11 and 7.12 of the Act - (section 7.4(3)(d))	The application of sections 7.11 and 7.12 of the Act is excluded in respect of the Development.		
Applicability of section 7.24 of the Act – (section 7.4(3)(d))	The application of section 7.24 of the Act is excluded in respect of the Development.		
Consideration of benefits under this deed if section 7.11 applies – (section 7.4(3)(e))	N/A		
Mechanism for Dispute Resolution – (section 7.4(3)(f))	See clause 8		
Enforcement of this deed – (section 7.4(3)(g))	See clause 6 and clause 7		
No obligation to grant consent or exercise functions – (section 7.4(10))	See clause 13.13		

Table 2 – Other matters

Requirement under the Act	This deed
Registration of the Planning Agreement – (section 7.6 of the Act)	Yes (see clause 7)
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before a Construction Certificate is issued – (clause 25E(2)(g) of the Regulation)	No
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before an Occupation Certificate is issued – (clause 25E(2)(g) of the Regulation)	No .
Whether the Planning Agreement specifies that certain requirements of the agreement must be complied with before a Subdivision Certificate is issued – (clause 25E(2)(g) of the Regulation)	No

Schedule 2 Address for Service

(clause 1.1)

Minister	
Contact:	The Secretary
Address:	Department of Planning, Industry and Environment 4 Parramatta Square 12 Darcy Street PARRAMATTA NSW 2150
Facsimile No:	Not applicable
Email:	planningagreements@planning.nsw.gov.au
Landowner and Developer	Hydro Aluminium Kurri Kurri Pty Ltd (ABN 55 093 266 221)
Contact:	The Directors
Address:	Hart Road, Loxford NSW 2326
Facsimile No:	Not applicable
Email:	kurri@hydro.com
Landowner and Developer's Legal Adviser	Gilbert + Tobin
Contact:	Practice Group Leader, Real Estate + Projects Group
Address:	Level 35, Tower 2, International Towers Sydney 200 Barangaroo Avenue, Barangaroo NSW 2000
Facsimile No:	(02) 9263 4111
Email:	info@gtlaw.com.au

Schedule 3 Land

(clause 1.1)

Lot	Deposited Plan	Folio Identifier	Landowner
Lot 3	456769	3/456769	Hydro Aluminium Kurri Kurri Pty Ltd
16	1082775	16/1082775	Hydro Aluminium Kurri Kurri Pty Ltd
318	755231	318/755231	Hydro Aluminium Kurri Kurri Pty Ltd
319	755231	319/755231	Hydro Aluminium Kurri Kurri Pty Ltd
411	755231	411/755231	Hydro Aluminium Kurri Kurri Pty Ltd
412	755231	412/755231	Hydro Aluminium Kurri Kurri Pty Ltd
413	755231	413/755231	Hydro Aluminium Kurri Kurri Pty Ltd
414	755231	414/755231	Hydro Aluminium Kurri Kurri Pty Ltd
420	755231	420/755231	Hydro Aluminium Kurri Kurri Pty Ltd
769	755231	769/755231	Hydro Aluminium Kurri Kurri Pty Ltd
1	456769	1/456769	Hydro Aluminium Kurri Kurri Pty Ltd
2	456769	2/456769	Hydro Aluminium Kurri Kurri Pty Ltd

Schedule 4 Development Contributions

(clause 4)

1. Development Contributions

(a) The Developer undertakes to provide the Development Contributions to the Minister in the manner set out in the table below:

Development Contribution	Timing	Value .
Monetary Contribution	In accordance with clause 2 of this Schedule 4	\$6,500,000
Containment Cell Contribution	In accordance with clause 3 of this Schedule 4	N/A
Remediation Works Contribution	In accordance with clause 3 of this Schedule 4	N/A

2. Monetary Contribution

- (a) The Developer must pay the Monetary Contribution to the Minister no later than 14 days from the Remediation Works Completion Date.
- (b) The Monetary Contribution will be an amount equal to the sum represented by "X" in the following formula:
 - (i) X = \$6,500,000 x (Current CPI/Base CPI)
- (c) To avoid doubt on each CPI Adjustment Date, the value of X in clause 2(b) will be adjusted by multiplying \$6,500,000 by an amount equal to the Current CPI divided by the Base CPI.
- (d) The parties agree that:
 - (i) the Monetary Contribution is to be not less than \$6,500,000; and
 - (ii) If the Monetary Contribution as adjusted in accordance with clause 2(b) is less than the amount of that contribution for the preceding 12 month period, then the Monetary Contribution continues to be the amount for that preceding 12 month period.

3. Remediation Works and Containment Cell Contribution

3.1 Developer's obligations

- (a) The Developer must:
 - (i) undertake the Remediation Works to the Minister's satisfaction and in accordance with the Remediation Consent and Schedule 6 of this deed;
 - (ii) maintain, manage and monitor the Containment Cell up to and during the Management Period in accordance with the LTEMP;

- (iii) provide the Access Road; and
- (iv) transfer the Containment Cell Land to the Minister free of cost, in accordance with clause 3.6 of this Schedule 4 of this deed.

3.2 Remediation Works Completion Date

- (a) The parties agree that for the purposes of this deed, the Remediation Works Completion Date is the date on which the Minister provides the Developer with a notice in accordance with clause 3.2(f).
- (b) Upon completion of the Remediation Works, the Developer must provide the Minister with a Remediation Validation Report.
- (c) The Remediation Validation Report must:
 - (i) be prepared in accordance with the Remediation Consent;
 - (ii) be addressed to the Minister or the Minister's nominee;
 - (iii) not identify any further works required in order for the Remediation Works to be achieved; and
 - (iv) otherwise be on terms satisfactory to the Minister or Minister's nominee (acting reasonably).
- (d) If the Remediation Validation Report states that:
 - (i) the remediation criteria set out in the RAP have not been achieved; and
 - (ii) additional site work is required to achieve these criteria (Additional Work),

the Developer must promptly undertake the Additional Work.

- (e) If the Developer considers that the Additional Work has been completed, the Developer will provide the Minister with a notice stating that the Additional Work has been completed, together with evidence that the Validation Consultant has confirmed that the Additional Work has been completed to the Validation Consultant's satisfaction (Validation Notice).
- (f) Upon receipt of a:
 - (i) Remediation Validation Report prepared in accordance with clause 3.2(c); or
 - (ii) Validation Notice,

the Minister will provide the Developer with a notice that the Remediation Works Completion Date has been achieved.

3.3 Management of the Containment Cell

- (a) The Developer must manage and maintain the Containment Cell up to and during the Management Period in accordance with the requirements of the LTEMP.
- (b) As soon as reasonably practicable after the date which is 5 years from the Remediation Works Completion Date, the Developer must, at its cost:
 - (i) obtain and provide to the Minister the Containment Cell Condition Report prepared by a Suitably Qualified Consultant which:

- (A) identifies whether any issues have arisen in relation to the Containment Cell during the Management Period which may present a risk to human health, safety or the environment; and
- (B) outlines any works which are to be undertaken in respect of the Containment Cell for the purposes of addressing any risks to human health, safety or the environment (**Rectification Works**).
- (c) The Developer must promptly undertake the Rectification Works if required and provide the Minister with a notice which states that the Rectification Works have been completed together with evidence to the reasonable satisfaction of the Minister that the Rectification Works have been attended to (**Rectification Works Notice**).
- (d) The Minister must within a reasonable period of receipt of:
 - a Containment Cell Condition Report which does not identify any Rectification
 Works, provide a written notice to the Developer which states that the Minister is satisfied with the Containment Cell Condition Report (acting reasonably); or
 - (ii) the Rectification Works Notice, provide a written notice to the Developer that:
 - (A) the Rectification Works have been completed to the Minister's satisfaction (acting reasonably); or
 - (B) the Rectification Works have not been completed to the Minister's satisfaction (acting reasonably).
- (e) Upon receipt of a notice under clause 3.3(d)(ii)(B), the Developer must provide evidence to the reasonable satisfaction of the Minister that the matters raised have been attended to and the provisions of clause 3.3(d)(ii) will again apply .
- (f) For the avoidance of doubt:
 - (i) any failure by the Minister to provide a notice in accordance with this clause 3.3(d) does not affect the Developer's obligation to transfer the Containment Cell Land to the Minister in accordance with clause 3.6 of Schedule 4 of this deed; however
 - (ii) if the Developer does not comply with the requirements of this clause 3.3, the Minister may refuse to accept the transfer of the Containment Cell Land.

3.4 Access Road

- (a) The Developer must provide the Access Road to the Containment Cell Land which provides the Minister with full and unfettered access over that Access Road from the nearest Public Road, being Hart Road, on and from the Containment Cell Land Transfer Date.
- (b) The Developer may, at any time prior to the Minister's approval of the Access Road Plan pursuant to clause 3.4(g) of this Schedule 4, notify the Minister in writing that it wishes to amend the Access Road Concept Plan.
- (c) The notice for the purposes of clause 3.4(b) must contain an amended Access Road Concept Plan, together with reasons why the Developer considers any such amendments are reasonably necessary to facilitate the redevelopment of the Land, and will not impede the Developer from complying with its overriding obligation under clause 3.4(a) of this Schedule 4, to provide the Access Road to the Containment Cell Land.

- (d) The Minister must, acting reasonably and within a reasonable period of receipt of a notice pursuant to clause 3.4(b) of this Schedule 4, provide a written notice to the Developer that it:
 - (i) approves the amended Access Road Concept Plan (Amended Access Road Concept Plan) in which case the Amended Access Road Concept Plan will become the Access Road Concept Plan for the purposes of this deed; or
 - (ii) does not approve the amended Access Road Concept Plan, together with the reasons why the amended Access Road Concept Plan is not approved.
- (e) Following the approval of the Amended Access Road Concept Plan in accordance with clause 3.4(d)(i) of this Schedule 4:
 - (i) the Minister agrees to do all things reasonably required by the Developer to release and discharge this deed with respect to any part of the Land that no longer forms part of the Access Road Land as a result of any Amended Access Road Concept Plan approved by the Minister pursuant to clause 3.4(d)(i) of this Schedule 4, subject to the requirements of clause 7.3 of this deed; and
 - (ii) the Developer must ensure that this deed is registered on that part of the Land which forms the Access Road Land as shown on the Amended Access Road Concept Plan approved by the Minister pursuant to clause 3.4(d)(i) of this Schedule 4 and in accordance with the requirements of clauses 7.1 and 7.2 of this deed.
- (f) Prior to the transfer of the Containment Cell Land and within 2 years of the Remediation Works Completion Date, the Developer must submit to the Minister for approval:
 - (i) a plan showing the proposed location of the Access Road; and
 - (ii) details of whether the Access Road is to be a Public Road or a Private Road,

(Access Road Plans).

- (g) The Minister must, acting reasonably and within a reasonable period of receipt of the Access Road Plans provide a written notice that he or she:
 - (i) approves the Access Road Plans; or
 - (ii) does not approve the Access Road Plans, together with the reasons why the Access Road Plans are not approved.
- (h) Upon receipt of a notice under clause 3.4(g)(ii), the Developer must provide evidence to the reasonable satisfaction of the Minister that the matters raised have been attended to and provide to the Minister (as appropriate) revised Access Road Plans.
- (i) The Minister agrees to do all things reasonably required by the Developer to release and discharge this deed with respect to any part of the Land that no longer forms part of the Access Road Land as a result of any Access Road Plan approved by the Minister pursuant to clause 3.4(g)(i) of this Schedule 4, subject to the requirements of clause 7.3 of this deed.
- (j) The Developer must ensure that this deed is registered on that part of the Land which forms the Access Road Land as a result of any Access Road Plan approved by the Minister pursuant to 3.4(g)(i) of this Schedule 4 and in accordance with the requirements of clauses 7.1 and 7.2 of this deed.
- (k) The Developer must, at its cost:

- (i) obtain all Approvals necessary for the construction of the Access Road; and
- (ii) construct the Access Road in accordance with:
 - (A) the Access Road Plans approved by the Minister under clauses 3.4(g) and (h);
 - (B) any Approvals for the Access Road;
 - (C) any relevant requirements of the Roads Authority;
 - (D) all applicable laws and standards; and
 - (E) good industry practice.
- (I) If the Developer considers that the Access Road has been completed, the Developer will forward to the Minister a written notice stating that Practical Completion has been achieved, together with:
 - (i) a copy of all Approvals for the Access Road;
 - (ii) to the extent that the Access Road is a Public Road, evidence:
 - (A) that the Access Road has been completed to the satisfaction of the Roads Authority and in accordance with clause 3.4(k)(ii); and
 - (B) that the Access Road has been dedicated to the Roads Authority; and/or
 - (iii) to the extent that the Access Road is a Private Road:
 - (A) an Access Road Certificate of Compliance; and
 - (B) a copy of the proposed dealing for the Access Easement (Proposed Dealing),

(Road Works Completion Notice).

- (m) The Minister must within a reasonable period of receipt of the Road Works Completion Notice, provide a written notice that states that the Minister is:
 - (i) satisfied (acting reasonably) that the Developer has complied with the applicable requirements of clause 3.4(l); or
 - (ii) not satisfied (acting reasonably) that the Developer has complied with the applicable requirements of clause 3.4(I).
- (n) As soon as reasonably practicable following receipt of a notice under clause 3.4(m)(ii), the Developer must provide evidence to the reasonable satisfaction of the Minister that the matters raised have been attended to and provide to the Minister (as appropriate) revised Road Works Completion Notice.
- (o) If the Developer does not comply with this clause 3.4, the Minister may:
 - (i) refuse to accept the transfer of the Containment Cell Land until such time that the Developer does comply with this clause 3.4; and
 - (ii) without limiting the Developer's obligations under clause 3.4(a), require that the Developer undertake works, at the Developer's cost and within a timeframe determined by the Minister (acting reasonably), so as to ensure that the Minister is provided with the Access Road.

3.5 Subdivision of the Containment Cell Land

- (a) In order to give effect to the transfer of the Containment Cell Land to the Minister in accordance with clause 3.6 of this Schedule 4:
 - (i) there must be a subdivision of the Land to create the parcel(s) of land that will comprise the Containment Cell Land; and
 - (ii) to the extent that the Access Road is a Private Road, the Access Easement must be registered on the Access Road Land.
- (b) The Developer must (at its cost) and prior to the Containment Cell Land Transfer Date, provide the Minister with a copy of the Containment Cell Land Subdivision Plan for approval (such approval not to be unreasonably withheld or delayed).
- (c) The Minister will provide written notice to the Developer stating whether the Containment Cell Subdivision Plan is approved within 20 Business Days of receipt of the Containment Cell Subdivision Plan. If a notice under this clause:
 - (i) states that the Containment Cell Subdivision Plan is approved, then it will be taken to be the approved Containment Cell Subdivision Plan; or
 - (ii) states that the Containment Cell Subdivision Plan is not approved and gives reasons why it is not approved or requests reasonable changes in order for the Containment Cell Subdivision Plan to be approved, the Developer must, within 20 Business Days of the Minister's notification under this clause, provide a revised Containment Cell Subdivision Plan in which case the provisions of clauses 3.5(a) to (c) of this Schedule 4 will continue to apply until the Containment Cell Subdivision Plan is approved by the Minister in accordance with clause 3.5(c)(i) of this Schedule 4.
- (d) Following the receipt of a notice from the Minister in accordance with clause 3.5(c)(i), the Developer must:
 - use all reasonable efforts to obtain Development Consent (if any is required) and any other approvals necessary to create a separate lot for the Containment Cell Land in accordance with the approved Containment Cell Subdivision Plan; and
 - (ii) in accordance with the applicable Development Consent (if any) and any other necessary approvals, register the Containment Cell Subdivision Plan to create a separate lot for the Containment Cell Land; and
 - (iii) attend to all steps necessary to register the Proposed Dealing, including:
 - (A) arrange for the Proposed Dealing to be signed by all required parties, including but not limited to the registered proprietor, mortgagee, chargee and/or covenant chargee of the burdened lot;
 - (B) obtain written consents to the registration of the Proposed Dealing from all required parties, including but not limited to any caveator, lessee and judgement creditor under any writ of the burdened lot; and
 - (C) lodge the Proposed Dealing for registration at the Land Registry Services NSW, and respond to and settle any requisitions raised.

3.6 Transfer of the Containment Cell Land

(a) On a date that is:

- (i) 3 months after the later of either:
 - (A) receipt by the Developer of a notice in accordance with clause 3.3(d)(i) of this Schedule 4; or
 - (B) receipt by the Developer of a notice from the Minister in accordance with clause 3.3(d)(ii)(A) of this Schedule 4; or
- (ii) within 3 months of receipt of a written request from the Minister,

the Developer agrees to deliver to the Minister:

- (iii) a form of transfer in respect of that part of the Land comprising the Containment
 Cell Land executed by the Developer and in registrable form;
- (iv) the certificates of title for the Containment Cell Land; and
- (v) if required, evidence that the Access Easement has been registered over the Access Road,

and must take any other necessary action to give effect to the transfer of the title of the Containment Cell Land to the Minister free of all encumbrances and affectations (including any charge or liability for rates, taxes and charges).

- (b) Upon transfer, the Containment Cell Land will be free from any encumbrances other than service easements or such other encumbrances as agreed with the Minister and such agreement by the Minister must not be unreasonably withheld or delayed.
- (c) The Developer must promptly comply, or procure compliance with, any requisitions raised by the Registrar-General in relation to the transfer of the Containment Cell Land which are within its power to comply with. The Minister must promptly comply with, or procure compliance with, any requisitions raised by the Registrar-General in relation to the transfer of the Containment Cell Land which are within the Minister's power to comply with.
- (d) The Developer will pay all rates and taxes owing in respect of the Containment Cell Land up to and including the date that the Developer delivers the form of transfer and certificates of title for the Containment Cell Land pursuant to clause 3.6(a) of this Schedule 4, after which time the Minister will be responsible for all rates and taxes in relation to the Containment Cell Land.
- (e) The Developer must, from the Commencement Date, not undertake any action or activity, prior to the Transfer being registered that will have a detrimental effect on the Containment Cell Land except where the Developer is:
 - (i) complying with the terms of this deed;
 - (ii) directed to undertake such action or activity by another government agency or instrumentality (such as the Rural Fire Service);
 - (iii) maintaining existing access and existing tracks; or
 - (iv) otherwise required by law to undertake such an action or activity including in order to comply with any conditions of a Development Consent or any order made under the Act in respect of the Land.
- (f) The Developer must, from the Containment Cell Land Transfer Date, assign any available warranties related to the construction, design, quality assurance, and liner materials with

respect to the Containment Cell to the Minister or (if the Developer is requested to do so by the Minister) the Minister's nominee, and provide all documentation relating to the design, construction and operation of, and material encapsulated in, the Containment Cell.

- (g) The Minister agrees, on and from the Containment Cell Land Transfer Date, to manage the Containment Cell Land in accordance with the requirements of the LTEMP.
- (h) The Developer indemnifies and agrees to keep indemnified the Minister against all Claims made against the Minister arising from or in connection with Contamination on, in or under or emanating from the Containment Cell Land, but only in relation to Contamination that existed on or before the Containment Cell Land Transfer Date.

3.7 Insurance

- (a) The Developer must take out and maintain, with a reputable insurer, insurances between the Remediation Works Completion Date and the Containment Cell Land Transfer Date (Containment Cell Insurance).
- (b) The Containment Cell Insurance must:
 - (i) cover the following items:
 - (A) failure of or damage to the Containment Cell;
 - (B) repair of the Containment Cell in the event of severe damage to or failure of the Containment Cell; and
 - (C) the costs of remedying any Contamination caused by the failure of or damage to the Containment Cell;
 - (ii) provide minimum coverage of \$10,000,000;
 - (iii) be on an occurrence, not claims made, basis;
 - (iv) name the "Minister for Planning and Public Spaces" and the "Department of Planning, Industry and Environment ABN 20 770 707 468" as an insured party, and the Minister's nominee if requested to do so by the Minister; and
 - (v) otherwise be on terms satisfactory to the Minister (acting reasonably).
- (c) On:
 - (i) or before the Remediation Works Completion Date, the Developer must provide to the Minister, a copy of a certificate of currency for the Containment Cell Insurance;
 - (ii) or before each anniversary of the Remediation Works Completion Date, the Developer must provide the Minister with a replacement certificate of currency for the Containment Cell Insurance to ensure that, at all times, until the Containment Cell Land Transfer Date, the Developer is in possession of the Containment Cell Insurance; and
 - (iii) for a period of 10 years immediately following the Containment Cell Land Transfer Date, the Developer must procure and fund the Containment Cell Insurance which must name the Minister and the Minister's nominee as an insured party.
- (d) The Developer must notify the Minister if:
 - (i) an insurance policy required by this clause 3.7 is cancelled;

- (ii) an event occurs which may give rise to a claim under, or which may affect rights under, an insurance policy in connection with the Containment Cell; or
- (iii) if the Developer becomes aware that the insurer is or may be insolvent.

3.8 Compulsory Acquisition

- (a) If the Developer does not transfer the Containment Cell Land in accordance with this deed, then without limiting any other rights or remedies which the Minister may have under this deed or otherwise, the Minister may elect that the Minister's nominee may compulsorily acquire the whole or any part of the Containment Cell Land in accordance with the Land Acquisition (Just Terms Compensation) Act 1991 (NSW) (Acquisition Act), for the amount of \$1.
- (b) The Developer agrees that:
 - (i) this clause 3.8 is an agreement between the Developer and the Minister's nominee, for the purposes of section 30 of the Acquisition Act;
 - (ii) in this clause 3.8, the Developer has agreed on all relevant matters concerning the compulsory acquisition and the compensation to be paid for the acquisition by the Minister's nominee.
- (c) If the Minister's nominee must pay compensation under Part 3 of the Acquisition Act to any person, other than the Developer, in accordance with the compulsory acquisition arrangements under this clause 3.8, the Developer:
 - (i) must reimburse the amount of that compensation to the Minister's nominee on request; and
 - (ii) indemnifies and agrees to keep indemnified the Minister's nominee against all claims made against the nominee pursuant to the Acquisition Act as a result of any acquisition by the nominee of the whole or any part of the Containment Cell Land under this clause 3.8.
- (d) The Developer must pay the Minister and the Minister's nominee, promptly on demand, an amount equal to all costs, charges or expenses incurred by the Minister and the nominee in relation to the acquisition of the whole or any part of the Containment Cell Land as contemplated by this clause 3.8.
- (e) The Minister agrees to manage the Containment Cell Land in accordance with the requirements of the LTEMP on and from the date the Minister acquires the Containment Cell Land in accordance with this clause 3.8.
- (f) The Developer indemnifies and agrees to keep indemnified the Minister's nominee against all Claims made against the nominee as a result of any Contamination that is required to be remediated by an Authority but only in relation to Contamination that existed on, in or under the Containment Cell Land on or before the date that the Containment Cell Land is acquired by the nominee whether under this clause 3.8 or any other clause of this Schedule 4.
- (g) The Developer acknowledges and agrees that this clause 3.8 operates as a deed poll in favour of the Minister's nominee.

Schedule 5 Security terms

(clause 6.2)

1. Developer to provide Security

- (a) In order to secure the payment or performance of the Development Contribution (other than the Remediation Works Contribution), the Developer has agreed to provide the Security in the form of Bank Guarantees as set out in the table below.
- (b) Each Bank Guarantee must:
 - (i) name the "Minister for Planning and Public Spaces" and the "Department of Planning, Industry and Environment ABN 20 770 707 468" as the relevant beneficiaries;
 - (ii) be in the amount as set out in the table below;
 - (iii) be as security for the Secured Obligation as set out in the table below; and

Bank Guarantee	Bank Guarantee Amount	Secured Obligation	Timing for provision of Security	
Monetary Contribution Bank Guarantee	\$6,500,000	The requirement to provide the Monetary Contribution	On the Commencement Date	
Initial Period Bank Guarantee	\$1,000,000	The requirements imposed on the Developer pursuant to clause 3 of Schedule 4 (other than with respect to the Remediation Works Contribution)	On the date that the Monetary Contribution Bank Guarantee is returned to the Developer	

(iv) not have an expiry date.

2. Claims under Bank Guarantees

- (a) The Minister may:
 - (i) call upon the relevant Security provided in accordance with this deed where the Developer:
 - (A) has failed to provide the Monetary Contribution on or before the date for payment under this deed; or
 - (B) has failed to comply with its obligations pursuant to clause 3 of Schedule 4; and
 - (ii) retain and apply such monies towards the Development Contribution and any costs and expenses incurred by the Minister in rectifying any default by the Developer under this deed.

- (b) Prior to calling upon the Security, the Minister must give the Developer not less than 10 Business Days written notice of his or her intention to call upon the Security and if the Developer remedies the breach or non-compliance to the Minister's satisfaction within that period the Minister will not call upon the Security.
- (c) If:
 - (i) the Minister calls upon a Security; and
 - (ii) applies all or part of such monies towards the Development Contribution and any costs and expenses incurred by the Minister in rectifying any default by the Developer under this deed; and
 - (iii) has notified the Developer of the call upon the Security in accordance with clause
 2(b) of this Schedule 5,

then the Developer must provide the Minister with a replacement Security to ensure that, at all times, until the date the Security is released in accordance with clause 3 of this Schedule 5, the Minister is in possession of Security for a face value equivalent to the relevant Security required to be provided in accordance with clause 1 of this Schedule 5.

3. Release of Security

3.1 Release of the Monetary Contribution Bank Guarantee

lf:

- the Developer has paid the Monetary Contribution and satisfied all of its obligations under this deed with regards to the Secured Obligation for the Monetary Contribution Bank Guarantee;
- (b) the whole of the Monetary Contribution Bank Guarantee has not been expended; and
- (c) the Developer has provided the Initial Period Bank Guarantee,

then the Minister will promptly return the Monetary Contribution Bank Guarantee (less, if applicable, any amounts properly claimed by the Minister from the Monetary Contribution Bank Guarantee under this deed) to the Developer.

3.2 Release of the Initial Period Bank Guarantee

lf:

- (a) the Developer has satisfied all of its obligations under this deed with regards to the Secured Obligation for the Initial Period Bank Guarantee; and
- (b) the whole of the Initial Period Bank Guarantee has not been expended,

then the Minister will on the Containment Cell Land Transfer Date promptly return the Initial Period Bank Guarantee (less, if applicable, any amounts properly claimed by the Minister from the Initial Period Bank Guarantee under this deed) to the Developer.

Schedule 6 Remediation Works

1. Remediation Works

- (a) The Developer must undertake (or cause to be undertaken on its behalf) the Remediation Works:
 - (i) in accordance with this deed;
 - (ii) in accordance with the Remediation Consent;
 - (iii) in accordance with the Scope of Works and so as to achieve the objectives of the RAP;
 - (iv) so that that the Remediation Works are Fit for Purpose;
 - so that the Remediation Works Completion Date occurs on or before the Date for Completion;
 - (vi) in compliance with all applicable laws and standards; and
 - (vii) in accordance with good industry practice.

2. Scope of Works

- (a) The parties agree that at the Commencement Date, the Scope of Works is the document attached at Schedule 7. The Developer is solely responsible for ensuring the Scope of Works meets the objectives of the RAP and satisfies the requirements of this deed.
- (b) The Developer must develop and execute the Scope of Works so that the Remediation Works:
 - (i) achieves the objectives of the RAP; and
 - (ii) otherwise complies with this deed.
- (c) As the Developer updates and amends the Scope of Works it must:
 - (i) to the extent reasonably practicable not alter the staging of the Remediation Works set out in the Scope of Works attached to this deed; and
 - (ii) provide the updated Scope of Works document to the Minister.
- (d) Should it be necessary to alter the staging of the Remediation Works the Developer must put a written proposal to do so to the Minister setting out the proposed changes and the reasons and the parties must negotiate in good faith to agree any required changes to the Security consequent upon the proposed changes. The Developer agrees that under no circumstances will the total amount of Security be reduced pursuant to the process under this clause 2(d).
- (e) If the Minister provides any comments on the Scope of Works the Developer must give due consideration to those comments.
(f) No act of omission on the part of the Minister in respect of the Scope of Works will derogate from the Developer's responsibility under clause 2(a).

3. Independent Engineer

- (a) Prior to the commencement Stage 1B Works and no later than 2 months from the commencement of the Remediation Works (**Required Date**), the Developer must, at its cost:
 - (i) engage an Independent Engineer to inspect and certify the Construction Remediation Works as being in accordance with:
 - (A) this deed;
 - (B) the Remediation Consent;
 - (C) all applicable laws and standards; and
 - (ii) cause the Independent Engineer to enter into the Independent Engineer's Deed.
- (b) The Developer acknowledges and agrees that if an Independent Engineer has not entered into the Independent Engineer's Deed by the Required Date, then the Remediation Works must cease and cannot be recommenced until such time as an Independent Engineer has entered into the Independent Engineer's Deed.
- (c) The Developer must perform (or procure the performance of) the Remediation Works such that the Independent Engineer can properly issue the certifications in clause 6 of this Schedule 6.

4. Developer to provide Security

- (a) In order to secure the performance of the Remediation Works the Developer has agreed to provide the Security in the form of Bank Guarantees as set out in the table below.
- (b) Each Bank Guarantee must:
 - (i) name the "Minister for Planning and Public Spaces" and the "Department of Planning, Industry and Environment ABN 20 770 707 468" as the relevant beneficiaries;
 - (ii) be in the amount as set out in the table below;
 - (iii) be as security for the Secured Obligation as set out in the table below; and
 - (iv) not have an expiry date.

Bank Guarantee	Bank Guarantee Amount	Secured Obligation	Timing for provision of Security
Remediation	\$4,000,000	Construction of Project Infrastructure	On the
Works Bank		as outlined in the Scope of Works	Commencement
Guarantee 1		(SP2 Part 1 Works)	Date

Remediation	\$8,000,000	Construction of the Containment Cell	On the
Works Bank		Stage 1 as outlined in the Scope of	Commencement
Guarantee 2		Works (SP2 Part 2 Works).	Date
Remediation	\$13,000,000	Performance of the Site Remediation	On the
Works Bank		& Material Transfer as outlined in the	Commencement
Guarantee 3		Scope of Works (SP2 Part 3 Works).	Date
Remediation Works Bank Guarantee 4	\$5,000,000	Construction of the Containment Cell Stage 2 & Completion as outlined in the Scope of Works (SP2 Part 4 Works) and compliance with Schedule 4.	On the Commencement Date

5. Extension of the Date for Completion

- (a) The Developer must proceed with the Remediation Works with due expedition and without unreasonable delay.
- (b) If the Developer is delayed in performance of the Remediation Works for reasons outside of its control, subject to the Developer's compliance with clause 5(a), the Minister may in his or her sole and unfettered discretion extend the Date for Completion by written notice to the Developer.

6. Certification and Completion of the Remediation Works

- (a) Each Stage of the Remediation Works will achieve Stage Completion when:
 - (i) the relevant Remediation Works are complete save for minor defects which:
 - (A) in no way impact the next Stage of works;
 - (B) do not prevent certification (or require some qualification on certification) of the relevant works; and
 - (C) are capable of being rectified during a subsequent Stage of works;
 - (ii) the documents indicated for the relevant stage in the table in clause 6(b) below have been provided to the Minister; and
 - (iii) any other document, record or thing reasonably required by the Minister has been provided to the Minister.
- (b) The certification and reporting documents required pursuant to clause 6(a)(ii) for each stage are as follows:

Stage No	Stage Description	Certification	Certification provided by
SP2 Part 1	Stage 1A Works	Stage 1A Certificate of Compliance	Validation Consultant
SP2 Part 1	Stage 1B Works	Certification in accordance with the Independent Engineer's Deed	Independent Engineer

SP2 Part 2	Containment Cell Stage 1 – Containment Cell Base	Certification in accordance with the Independent Engineer's Deed	Independent Engineer
SP2 Part 3	Site Remediation & Material Transfer	Interim Statement	Site Auditor
SP2 Part 4	Containment Cell Stage 2 – Containment Cell Capping and Completion	Certification in accordance with the Independent Engineer's Deed Site Audit Statement and Site Audit Report	Independent Engineer Site Auditor

7. Claims under Bank Guarantees

- (a) The parties acknowledge and agree that the Bank Guarantees are provided by the Developer as security for the Developer's performance of its obligations under this deed.
- (b) The Minister may call upon any or all of the Bank Guarantees (as may be remaining in the Minister's possession at the time) if the Remediation Works Completion Date has not occurred by the Date for Completion.
- (c) The Minister may retain and apply such monies towards any costs and expenses incurred by the Minister in rectifying any default by the Developer under this deed.
- (d) Without limiting clause 7(b), regardless of whether a dispute exists regarding the circumstances in which a demand on the Bank Guarantees can be made, the Minister may make demand upon any Bank Guarantee provided pursuant to this deed and use the proceeds of such demand:
 - to recover any damage, loss, liability or cost (of any description) suffered or incurred by the Minister, or which the Minister reasonably considers he or she will suffer or incur, under or in connection with this deed for which the Developer is, or will be, liable under or in connection with this deed;
 - (ii) to recover any moneys or debt due from the Developer to the Minister;
 - (iii) in respect of any bona fide claim made by the Minister against the Developer under or in connection with the relevant Secured Obligation;
 - (iv) where this deed is terminated by the Minister due to the default of the Developer; or
 - (v) where the Developer is insolvent, being wound up (voluntarily or otherwise) or experiencing financial distress.
- (e) Prior to calling upon the Security, the Minister must give the Developer not less than 10 Business Days written notice of his or her intention to call upon the Security and if the Developer remedies the breach or non-compliance to the Minister's satisfaction within that period the Minister will not call upon the Security.

- (f) If:
 - (i) the Minister calls upon a Security; and
 - (ii) applies all or part of such monies towards the costs and expenses incurred by the Minister in rectifying any default by the Developer under this deed; and
 - (iii) has notified the Developer of the call upon the Security in accordance with clause 7(e) of this Schedule 6,

then the Developer must provide the Minister with a replacement Security to ensure that, at all times, until the date the Security is released in accordance with clause 8 of this Schedule 6, the Minister is in possession of Security for a face value equivalent to the relevant Security required to be provided in accordance with clause 4 of this Schedule 6.

- (g) Regardless of whether a dispute exists regarding the circumstances in which a demand on a Bank Guarantee can be made, the Developer must not take any steps to prevent or restrain:
 - (i) the Minister making a demand on the Bank Guarantees;
 - (ii) the issuer of a Bank Guarantee from making a payment under the Bank Guarantee; or
 - (iii) the Minister using any proceeds of any demand on a Bank Guarantee.
- (h) Where the Minister has converted a Bank Guarantee into money, any interest earned on such monies shall be retained by the Minister.
- (i) The Minister is not obliged to hold amounts realised on the demand on a Bank Guarantee in any definable account.
- (j) The Minister does not hold amounts realised on the demand on a Bank Guarantee on trust for the Developer and the Developer does not have ownership of, or any proprietary interest in, such amounts.

8. Release of Security

8.1 Release of the Remediation Works Bank Guarantee 1

Unless the whole of the Remediation Works Bank Guarantee 1 has been expended, the Minister must promptly return the Remediation Works Bank Guarantee 1 after the SP2 Part 1 Works reach Stage Completion.

8.2 Release of the Remediation Works Bank Guarantee 2

Unless the whole of the Remediation Works Bank Guarantee 2 has been expended, the Minister must promptly return the Remediation Works Bank Guarantee 2 after the SP2 Part 2 Works reach Stage Completion.

8.3 Release of the Remediation Works Bank Guarantee 3

Unless the whole of the Remediation Works Bank Guarantee 3 has been expended, the Minister must promptly return the Remediation Works Bank Guarantee 3 after the SP2 Part 3 Works reach Stage Completion.

8.4 Release of the Remediation Construction Works Bank Guarantee 4

Unless the whole of the Remediation Works Bank Guarantee 4 has been expended, the Minister must promptly return the Remediation Works Bank Guarantee 4 after the:

- (a) requirements for completion of the Remediation Works set out in the Remediation Consent have been met to the reasonable satisfaction of the Minister; and
- (b) Developer has complied with the requirements of clause 3.3 and clause 3.4 of Schedule 4.

9. Right of entry

- (a) If the Developer is in breach of any of its obligations under Schedule 4, Schedule 5 or Schedule 6 of this deed, and has failed to rectify any such breach within a reasonable period of time:
 - (i) the Minister may, at any time, in its absolute discretion:
 - (A) take exclusive or non-exclusive possession of the whole or part of the Land, at no fee; and
 - (B) elect any person including but not limited to a contractor, agent or employee of the Minister to carry out any work in connection with the Remediation Works or any other works relating to the Land that the Minister determines are required (acting reasonably), on the Developer's behalf, including, for the avoidance of doubt, any works during the Management Period. Without limiting the Minister's right to call on the Securities referred to in Schedule 5 and Schedule 6, the Developer must promptly reimburse the Minister for all additional costs incurred in respect of this clause 9(a)(i)(B) if the Security is not sufficient to cover the costs incurred by the Minister, or costs the Minister reasonably expects to incur;
 - (ii) the Developer indemnifies the Minister against any liability and costs arising from or incurred in connection with the Minister exercising its rights to enter and undertake the Remediation Works or any other works required to be undertaken under clause 9(a)(i)(B) of this Schedule (including legal costs and expenses on a full indemnity basis or solicitor and own client basis, whichever is the higher);
 - (iii) for the purposes of clause 9(a)(i), the Developer must provide to the Minister all keys and other security related devices to allow the Minister (or any person appointed by the Minister) to access the Land;
 - (iv) the parties agree and acknowledge that the Minister may provide any third party with a copy of this clause 9 or this deed as evidence of the Minister's right to occupy and carry out activities on the Land; and
 - (v) upon the Minister's request, the Developer must provide to the Minister copies of any material or information to which the Developer has access and which the Minister may reasonably require, including but not limited to:
 - (A) applications made or Approvals obtained by the Developer relating to the Remediation Works;
 - (B) any contamination and environmental reports prepared by or for the Developer in respect of the Remediation Works; and

- (C) notices from Authorities, neighbouring owners or other parties relating in any way to the Land.
- (b) For the avoidance of doubt, the Minister is not obliged to comply with the requirements imposed on the Developer under Schedule 4, Schedule 5 or Schedule 6 (such as taking out the Containment Cell Insurance or engaging an Independent Engineer).
- (c) The parties acknowledge and agree that any rights exercised by the Minister under clauses 7 and 9 of this Schedule 6 do not in any way reduce or limit the Developer's liability in respect of any breach of its obligations under Schedule 4, Schedule 5 or Schedule 6 of this deed.

in in

Schedule 7 Scope of Works





Hydro Aluminium Kurri Kurri Pty Ltd (ABN 55 093 266 221) Hart Rd, Loxford New South Wales 2327 Australia

Telephone: +61 2 4937 1555 Facsimile: +61 2 4937 4595

Web: www.hydro.com

Project:

HAKK Demolition and Remediation Project

Package:

Containment Cell Construction & Site Remediation

CONTRACT SR2018-001 Annexure Part H

Document:

Scope of Work Section 1 of 6 Overview

12th September 2019 Revision 0





			DOCUMENT CO	ONTROL	1	
			Signatures:			
0	12.09.2019	IFC	FOR CONTRACT	NH	AW	LP
Rev No	Date	Status	Revision Details	Originator	Verifler	Approver

1. This document and the information contained herein are strictly confidential.

2. This document is not suitable and/or approved for use unless signed above by the Approver.

3. A person must not use, ratain and/or reproduce this document unless authorised to do so by Hydro Aluminium Kurri Kurri Pty Ltd ("HAKK") and, if so authorised, must only use the document for the purpose that such authorisation was given.

4. A person using this document, including any information, data and/or documents contained or referenced herein, accepts the risk of:

 using the documents, information and/or data in electronic form without requesting and checking them for accuracy against the original hard copy version; and

b. using the documents, information and/or data for any purpose not agreed to in writing by HAKK.





Table of Contents

Α.	USING THE SCOPE OF WORK DOCUMENTS	6
1.	INTRODUCTION	8
2.	SITE AND LOCATION	8
3.	GENERAL	9
4.	DETERMINATION OF ALL MATERIAL QUANTITY	9
5.	METHODS OF MEASUREMENT OF QUANTITIES	.10
6.	TIMING OF THE PROJECT APPROVAL CONDITIONS	.10
7.	HIERARCHY OF APPROVALS	
8.	THE WORKS OVERALL	.11
9.	ENVIRONMENTAL CONTROLS	.15
10.	HOURS OF WORK	.15
11.	QUALITY ASSURANCE REQUIREMENTS	.15
11.1	. Superintendent's Construction Quality Assurance Engineer	. 16
11.2	. Contractor's Quality Assurance Representative	16
11.3		
11.4		
11.5	. Works As-Executed (WAE) DOCUMENTS	17
12.	SURVEY & SET OUT OF THE WORKS	.17
13.	EXISTING SERVICES	.18
14.	MATERIAL AND RESOURCES TO BE SUPPLIED BY THE PRINCIPAL	.18
15.	WORKS EXCLUDED	.18





Acronyms and Glossary

Acronym	Term or Phrase	Definition
of glossaries	s provided in other published documents s and the Technical Specification. Where	nts that form the Contract and are in addition to and not a substitutior s, which include the General Conditions of Contract, the Remedial any conflict exists, a clarification shall be sought from and provided
AEC	Area of Environmental Concern	Refer to Remedial Action Plans, Appendix 3 - Areas of the Site that require surface soil and sediment remediation
ВСМ	Bank Cubic Metres	The volume of material in its preserved, in-situ and undisturbed condition
ССМ	Compacted Cubic Metres	The volume of material following the placement and compaction of that material in its specified location
RWEMP	Remediation Works Environment Management Plan	The plan that provides the methods and means of the protection of the environment during the remediation and construction works
CQA	Construction Quality Assurance	Quality assurance relating to the construction works
CWS	Capped Waste Stockpile	The stockpile of waste material located at the Site designated as AEC1 in the Remedial Action Plan AS130349
ECC	Engineered Containment Cell	The structure designed for the containment of materials
EPA	Environment Protection Authority	Government authority relating to the protection of the environment
ITF	Independent Testing Firm	A company engaged for the performance of testing and inspections that has no commercial connection to or interest in the Works or any Suppliers, subcontractors or Contractors participating in the Works
PAEC	Potential Area of Environmental Concern	Refer to Remedial Action Plans, Appendix 3 - Areas of the Site that may require surface soil and sediment remediation
RAP	Remedial Action Plan	Document attached to the Scope of Works at Appendix 3 - Remediation strategy for soils at the Site published by Ramboll Australia
RFI	Request for Information	A formal question to the Superintendent from the Contractor, as described at Section 4 of Separable Portion SP1
SAS	Site Audit Statement	The statement written by the Site Auditor in relation to the Authorities' Consent Conditions
	Additive Material	Material that is contained in, on or under the Capped Waste Stockpile that is required to have Gypsum added prior to its placement in the ECC
	Approval Conditions Conditions of Consent Project Consent Conditions	The explicit conditions upon which the Project Approval is reliant and the characteristics that the Works must exhibit as a condition of their acceptance by the Authorities
	Authorities	Organisations that provide laws, ordinances, regulations, by- laws, and statutory powers that govern the Works
	Base Liner	The layers that are designed to be constructed underneath the ECC upon which material is to be placed
	Capping	The layers that are designed to be placed on top of the material placed within the ECC
	Contract	As defined in the General Conditions of Contract Clause 2 Interpretation
	Contract Documents	The documents that combine to represent the entire agreement between the parties and describe the Works





CQA Engineer	Organisation appointed by the Superintendent as his representative to carry out Quality Assurance tasks
 Development Consent	The details of the approval of the works by the Authorities
Footprint	The area underneath a stockpile or area containing material that is required to be relocated or removed
Impacted Material	Material on or under the Capped Waste Stockpile that is in direct contact with the waste material stored within the Capped Waste Stockpile
Jagged Material	Material exhibiting characteristics that present a risk of damage to the ECC Liner
Leachate Leachate Liquid	The fluid that is generated by water coming into contact with or percolating through Fluoride contaminated waste material
Miscellaneous Contaminated Materials	Material at the Site that is contaminated and specified to be placed in the ECC
Principal's Environment Management Plan	The published Hydro Aluminium Kurri Kurri Environment Management Plan, attached at Appendix 5 to the Scope of Works and as amended
Project	The Works described by the SR2018-001 Contract
Project Approval	The formal approval by the Authorities allowing the Works to be undertaken
Project Data Report	The document that provides formal evidence of compliance of the Works with the Contract and the Project Consent Conditions
Scope Scope of Work	The documents that combine to describe the Works that are to be carried out and completed in accordance with the Contract
Sidewall Liner	The layers that are designed to be constructed around the sides of the ECC upon which material is to be placed
Site Auditor	A person authorised by the Department of Planning & Environment to audit the Site to ascertain the Project's alignment with Project Consent Conditions
Special Material	Material classification for materials used in the construction of the ECC, that includes specific types as listed at Scope of Work Section 2 of 6, Section 7.3.3
Specification	The Containment Cell Design Technical Specification,
Technical Specification	reference 22/18015, attached as Appendix 2 to the Scope of Works and as amended
Temporary Accessway	An existing or new road, track or path used to gain access to areas of the Works during the period of the Works
Temporary Stockpile	An area allocated by the Superintendent for the storage of items and materials during the period of the Works and for their duration





A. USING THE SCOPE OF WORK DOCUMENTS

The Scope of Work comprises the documents listed in the table set out in this section A.

When reading, interpreting and using the Scope of Work (including its attachments), the following shall apply:

- References to the Contract will be taken to mean the Contract to which the Scope of Work is attached.
- To the extent there is any inconsistency between the Scope of Work and the General Conditions of the Contract, the General Conditions of the Contract will prevail.
- Terms used in the Scope of Work, which are defined in the General Conditions of the Contract, will have the meaning given to them in the General Conditions of the Contract. Terms which are used in the Scope of Work, which are not defined in the General Conditions of the Contract, but which are defined in any part of the Scope of Work, will have the meaning given to them in the Scope of Work.
- Unless the Scope of Work expressly states that an activity, task, item or obligation is to be supplied or carried out by the Principal, all activities, tasks, items and obligations are to be supplied or carried out by the Contractor at its own cost.
- The Scope of Work will be read and interpreted in accordance with the rules of interpretation set out in clauses 1 and 2 of the General Conditions of the Contract and, to the extent the following is not inconsistent with clauses 1 and 2 of the General Conditions of the Contract:
 - Where measurements or quantities are stated, the measurements and quantities shall be interpreted as approximate only and are not exact unless expressly stated as being 'exact'. The Contractor shall make all allowances considering the approximate nature of measurements and quantities.
 - For the majority of materials' volumetric quantities, in metric cubic metres (m³), the quantities given are from in-situ measurements and volumes are calculated from these measurements. The quantities used in this Scope of Works, prior to their movement from their stored location, is "Bank Cubic Metre", being a cubic metre of material in-situ and before being moved. The acronym "BCM" is used and not m³.
 - The quantities used in this Scope of Works, following their placement and compaction in the Engineered Containment Cell, is "Compacted Cubic Metre", being a cubic metre of material placed and compacted in its final location. The acronym "CCM" is used and not m3.
 - For materials' weight quantities, in metric tonnes, an estimate of 1.8 tonne per BCM has been used and is an estimate only. In-situ materials may be more or less than 1.8 tonnes per BCM.
 - The Contractor shall calculate their own volumes and tonnage for the purposes of materials handling (Loose Cubic Metres), and for adding Gypsum (Loose tonnes).
- The words 'include', 'includes' and 'including' shall not be limiting.
- 'Section', 'Appendix', and 'Attachment' refers to a section, appendix or attachment of this suite of Scope documents.
- Where the Scope of Work document (including its attachments) contains representations, the Contractor must not rely upon those representations and must





verify the correctness, accuracy and quality of information contained in such representations using its own knowledge, investigations and expert skill and judgement.

 Where a requirement is stated, the words used to describe that requirement shall not be interpreted as strictly limiting and the Contractor shall be responsible for carrying out all work reasonably necessary to achieve the requirement, including work that can be reasonably inferred as necessary to achieve that requirement.

This document is Section 1 of a 6-Section suite of documents that constitutes the Scope of Work, and which is included in the Contract. The documents that comprise the Scope of Work are listed below:

Order of Precedence	Scope of Work document
1	Scope of Work Section 1 of 6: Overview (this document)
2	Scope of Work Section 2 of 6: SP1 Preliminaries, Approvals & Temporary Works
3	Scope of Work Section 3 of 6: SP2 Part 1 Project Infrastructure
4	Scope of Work Section 4 of 6: SP2 Part 2 Containment Cell Stage 1
5	Scope of Work Section 5 of 6: SP2 Part 3 Site Remediation & Material Transfer
6	Scope of Work Section 6 of 6: SP2 Part 4 Containment Cell Stage 2 & Completion
7	Appendix 1 Technical Specification
8	Appendix 2 Drawings
9	Appendix 3 Remedial Action Plans
10	Appendix 4 NOT USED
11	Appendix 5 Miscellaneous Information
12	Appendix 6 Safety Data Sheets
13	Appendix 7 Site Reference Drawings
14	Appendix 8 Design Reports
15	Appendix 9 Site Remediation Area Handover Plan
16	Appendix 10 Site Photographs





1. INTRODUCTION

The Kurri Kurri Aluminium Smelter has operated at Hart Road Loxford, New South Wales since commissioning by Alcan in 1969. ("HAKK Smelter" or "Smelter" used interchangeably). The Smelter includes a plant area of approximately 60 hectares, contained within a 2,000-hectare buffer zone. Hydro Aluminium Kurri Kurri Pty Limited ("HAKK" or "Hydro" or "the Principal" used interchangeably) commenced ownership of the facility in 2002.

Smelting activities ceased at the site in September 2012, and in May 2014 the Principal formally announced the closure of the smelter.

It is the Principal's strategic vision for the Hydro land to play a key role in allowing the Hunter Region to achieve the economic, employment and environmental objectives identified in the NSW Government NSW State Plan 2021 and the Hunter Regional Plan 2036. Hydro aims to achieve this strategic vision by facilitating the rezoning and development of the Project site for significant employment, residential, rural and biodiversity conservation purposes.

The Principal has completed a number of decommissioning activities and is now progressing with the demolition and remediation of the Smelter area.

The Principal has engaged Ramboll Australia Pty Ltd as environment consultant for the whole of the Works, and GHD Pty Ltd for the design and specification of the Engineered Containment Cell forming part of the Works ("Engineered Containment Cell, "Containment Cell" or "ECC" used interchangeably).

2. Site and Location

The HAKK Smelter is in Loxford, NSW which is approximately 35km west of Newcastle and 150km north of Sydney.



Figure 2-1: HAKK Locality plan (insert: aerial of smelter plant vicinity)





The Smelter area is located on parts of Lot 3 of Deposited Plan (DP) 456769, Lot 16 DP1082775 Pt 1, Lot 411 DP755231, Lot 412 DP755231, Lot 413 DP755231, Lot 414 DP755231, Lot 420 of DP755231, Lot 318 DP755231, Lot 769 DP755231, Lot 319 DP755231 and Lot 1 DP456769.

Key roads in the vicinity of the Site are:

- Hart Road, which is immediately adjacent to the western section of the site and connects the Site to the M15 Hunter Expressway and the townships of Weston, Kurri Kurri and Cessnock.
- Dickson Road, which intersects with Hart Road approximately 120 metres south of the site security gate and is immediately adjacent to the eastern section of the site.
- M15 Hunter Expressway, which is approximately 380 metres to the southwest of the Site and connects the site to Sydney via the M1 Expressway to the South, or to Brisbane via the M1 Expressway or the New England Highway to the North.

3. General

The purpose of the Works is to clear waste materials, including contaminated materials, or Areas of Environmental Concern ("AEC"), from the Site for the Site to be subsequently utilised for future development.

To achieve this purpose the Contract will facilitate the construction of an ECC and the collection and relocation of materials from the Site, for the long-term storage of materials in the ECC. Subsequently, the Site will be audited the Principal's Site Auditor.

1	2	SCOPE OF WO	4	5	6
	SP1		\$f	2	
Overview	Preliminaries Approvals & Temporary Works	Part 1 Project Infrastructure	Part 2 Containment Cell Stage 1	Part 3 Site Remediation & Material Transfer	Part 4 Containmer Cell Stage 2 & Completion

The works have been divided into 6 (six) sections as follows:

4. Determination of all Material Quantity

The determination of all quantities of materials shall be by measured weight in tonnes. The Contractor shall use the Principal's Site weighbridge, or any other weighing asset provided by the Contractor that has been calibrated by an independent, NATA-certified, third party.

Payment for works based on the movement of materials shall be by weighed measurement and not dimensional survey, unless explicitly approved in writing by the Superintendent.





5. METHODS OF MEASUREMENT OF QUANTITIES

The works are predominantly characterised by the excavation, loading, haulage, and placement of bulk materials. As such the methods of measurement of the works shall be via the following means:

- A. Weight via the Contractor's NATA certified weighing asset, or the Hydro Aluminium Weighbridge (if this is available), which will provide each truck a certified gross and tare weight.
- B. Volume Bank Cubic Metres by in-situ survey prior to materials being removed from their undisturbed location, carried out by persons and using methods approved by the Superintendent for the performance of this work.
- C. Volume Compacted Cubic Metres by in-situ survey following final placement and compaction of materials, in accordance with the Technical Specification, at their specified final location, carried out by persons and using methods approved by the Superintendent for the performance of this work.
- D. Time Standby Time, Working Time and Non-Working Time as defined elsewhere in the Contract
- E. Dimension by use of a calibrated measuring device.

All other measurements shall be provided in accordance with AS 1181-1982 Methods of Measurement of Civil Engineering Works and Associated Building Works.

6. Timing of the Project Approval Conditions

The Contractor shall note that this project, the Containment Cell Construction and Site Remediation, is awaiting approval from the Authorities and, as such, no Separable Portion 2 works may proceed in any areas until that approval has been received and directed by the Superintendent.

The Project Approval from the Authorities will include the issue of Development Consent Conditions which will affect the Works. A Draft version of the Development Consent Conditions is included within the Contract.

The Works will comprise 2 distinct Separable Portions, being:

- Separable Portion 1 Preliminaries, Approvals and Temporary Works, which includes the works that are required to be carried out as part of the overall project and do not require Development Consent Approval in order for them to be undertaken and completed; and
- Separable Portion 2 The Balance of the Works, all of which requires Development Consent Approval

Upon receipt of the Development Consent Approval the specific Development Consent Conditions will form part of the Contract for these works and will be introduced prior to the commencement of the Works the subject of Separable Portion 2.

7. Hierarchy of Approvals

The Works are subject to approvals at each step of their progress, from a Contractor's draft of a management plan to the acceptance by a third-party authority, being the Site Auditor, for the final clearance of the whole of the Works.





The Superintendent's approval is referenced throughout the Contract where such approval is required.

The completion of the Works, and areas of the Works, is subject to the Superintendent's approval and, in specific circumstances, to the approval by other parties. These other parties' approvals include, but are not limited to, the following:

- Principal's Environmental Consultant's Approval Area environmental clearance certification, including asbestos clearance, who will provide a Validation Report for the Works;
- The Independent EPA-Accredited Site Auditor who will provide a Site Audit Statement ("SAS") for the Works
- Department of Planning and Environment overall project approval following their receipt of the SAS.

8. The Works Overall

It is estimated that an amount of approximately 600,000t of material, plus (subject to Final Development Consent Conditions) approximately 27,000t of Gypsum additive, is to be placed in the ECC that is to be constructed as part of the Works.



Except to the extent that the Contract otherwise expressly provides, the Contractor shall supply everything necessary for the proper performance of the Contractor's obligations and the discharge of the Contractors' liabilities under the Contract. This shall include suitable technical resources for labour, supervision, inspection, and project management, and the





provision of all materials, plant, equipment, supplies, temporary works and consumables that are required to safely execute the works.

The Works overall, including the RAP's, the drawings, the Specification and the Appendices, are defined in two Separable Portions that are generally described as follows:

- Separable Portion 1 (SP1) Preliminaries, Approvals & Temporary Works
 - o the preparation, issue and approval of management plans for the Works;
 - the preparation, issue and approval of a construction program for the Works, incorporating the approved management plans in so far as they affect the planned progress of the Works;
 - the completion of pre-qualification submissions for the materials specified for incorporation into the Works;
 - o the submission of requests for approval of suppliers for specified materials;
 - the submission of requests for the approval for the engagement of subcontractors for the Works, including liner installation and liner integrity testing;
 - the completion of detailed Works Method Statements for the ECC and for each specific element of the Works. The Superintendent's approval of these plans shall be provided in support of the Contractor's approved program;
 - the provision of Site access by the Principal for the Contractor's inspection and planning for their temporary facilities;
 - the creation of Temporary Stockpile areas for the collection of 'clean' materials, provided these areas do not require removal of native vegetation or disturbance of Aboriginal heritage items;
 - the creation of accessways to all Temporary Stockpiles and waste material stockpiles within the Smelter footprint, provided that no 'new' roads are included;
 - o the relocation of existing stockpiles to new Temporary Stockpiles for later use;
 - the installation of a temporary truck-weighing facility, at or near the Capped Waste Stockpile (CWS), for the weighing of project vehicles and the determination of material transported to the ECC, and the determination of Gypsum Additive quantities required; and
 - the completion of temporary soil and water management infrastructure, including environmental controls, around the Site excluding the area surrounding the ECC.
- Separable Portion 2 (SP2) Cell Construction and Remediation Works, following the procurement of the relevant approvals from the Superintendent from Separable Portion 1, these Separable Portion 2 works includes the construction and remediation works for the project including but not necessarily limited to, the following:

SP2 Part 1 Project Infrastructure including:

- o the setting up of the Contractor's temporary project facilities;
- o the creation of the main east-west haul road;
- o the installation of temporary fencing around all works' areas;
- the installation of Soil and Water Management infrastructure, including environmental protection provisions, at the ECC area;
- the clearing and excavation of the clay borrow pit area of the Site for ECC construction;





- the construction of a permanent creek crossing using culverts;
- the construction of 5 dams 1 Leachate dams and 3 sediment basins at the ECC Site and 1 Leachate Dam near the CWS;
- o the stockpiling of excavated materials to Temporary Stockpiles for later use;

SP2 Part 2 Containment Cell Stage 1 including:

- o the completion of any remaining Temporary Works
- the ECC Access Road
- the ECC Perimeter Road
- the first stage of the ECC construction including:
 - the installation of pipework for the collection and distribution of Groundwater underneath the ECC
 - installation of the 'Base Liner'
 - installation of separate pipework systems, within specific layers of the Liner, for the collection and distribution of Leachate Liquid and water from the ECC
 - the construction of sub-cells and access ramps
 - installation of the water and Leachate Liquid collection sumps
 - the installation of the 'Sidewall Liner' and Anchor trenches

SP2 Part 3 Site Remediation & Material Transfer including:

- the collection and determination of the weight of each load of material, approximately 600,000 tonnes, from identified Areas of Environmental Concern ("AEC's") at various locations around the Site that are referenced in the Remedial Action Plans ("RAP"), and its loading into the ECC in a prescribed sequence. The material includes, but is not limited to, the following:
 - Dickson Road Stockpile material [AEC29]
 - Asbestos Contaminated Material ("ACM") stored on Site in stockpiles at various locations
 - Capped Waste Stockpile ("CWS") [AEC1]
 - Miscellaneous Contaminated Materials
 - Process waste material stored at the 7A Furnace building north & south tubs
 - > General waste material from the demolition works
 - Anode Waste Pile [AEC2], and East and West Surge Ponds & associated drainage lines[AEC5,6,15]
 - Carbon Plant footprint [AEC8]
 - Bake Furnace Scrubber footprint [AEC26]
 - Dickson Road South material [RAP AS130419]





- the handling and addition of Gypsum to each load of material taken from the CWS, measured by weight per tonne of each load. All loads are to be weighed by the Contractor and the respective quantity of Gypsum material added to that load. The Gypsum will be free-issued and stored in a stockpile near the CWS. The Contractor will load the Gypsum as required into each load of CWS material. It is estimated (subject to Final Development Consent Conditions) that an amount of approximately 27,000 tonnes of Gypsum is to be supplied, by the Principal, and added, by the Contractor, to 270,000T of CWS Material;
- the installation of the first part of the Gas Management System within the upper level of waste material placed within the Cell;
- o the completion of the Leachate Transfer System (from Part 3);
- the procurement of Area Clearance Certificates, from Ramboll, for all areas of the Site that have been cleared of materials;
- following their certified clearance, the remediation of the various areas of the site by backfilling and resurfacing using clean fill to provide a free-draining site.

SP2 Part 4 Containment Cell Stage 2 & Completion including:

- o the placement of the capping layer of the Cell;
- o the completion of the Gas Management System (from Part 4);
- o the completion of the vegetation layer;
- the certification of the ECC Works by the CQA Engineer;
- the removal of all Temporary Works and reinstatement of the affected landform, including Leachate storage dams
- the completion of the Works according to all statutory processes and EPA requirements such that:
 - Ramboll will provide a Validation Report to that effect; and
 - the Site Auditor will provide a Site Audit Statement to that effect;
- The provision of a Project Data Report from the Contractor containing all documents that are required to be developed, produced, provided and procured by the Contractor during and following the works as required to be provided in accordance with the Contract; and





 The satisfaction of all requirements for the qualification for the award of Practical Completion by the Superintendent of the whole of the Works.

9. Environmental Controls

The Works shall be under the governance of Development Consent Conditions that include specific environmental controls. These controls shall include, but will not necessarily be limited to, the following:

- Air quality impact controls
- Dust minimisation controls
- Odour minimisation controls
- Noise minimisation controls
- Erosion and sediment controls
- Surface water and groundwater controls
- Traffic controls
- Spill response
- Handling of hazardous materials
- Handling of waste materials
- Protection of flora and fauna

The Contractor shall develop their Construction Environment Management Plan as part of Separable Portion 1 Preliminaries & Approvals that shall incorporate the requirements of the RAP, the Principal's Environment Management Plan and the Development Consent Conditions.

10. Hours of Work

Subject to Final Consent Conditions, the Contractor shall only undertake works associated with the Works that may generate an audible noise at the closest residential receptor during the following hours unless under direction from a relevant authority for safety reasons or in the event of an emergency:

- 7.00 am to 6.00 pm, Monday to Friday;
- 7.00 am to 1:00 pm on Saturdays; and
- At no time on Sundays or public holidays.

11. Quality Assurance Requirements

The Technical Specification, attached at Annexure 1 ("Specification"), provides details for quality assurance for the Works, and includes details in relation to inspection, testing, certification, verification and acceptance of all elements of the works.

Table 1-3 in the Specification provides a list of Witness Points and Hold Points required to be incorporated into the Contractor's Quality Assurance Plan (Specification section 1.12 refers).

The Contractor shall incorporate the details of quality assurance activities and actions in their work method statements and shall be responsible for the complete satisfaction of the technical specification's provisions, including the provision of documentary evidence of that complete satisfaction.





The Specification includes descriptions of Acceptance Criteria, including acceptance criteria in tabular format, for all measurable elements of the ECC Works.

11.1. Superintendent's Construction Quality Assurance Engineer

The Superintendent shall appoint a Construction Quality Assurance Engineer ("CQA Engineer") to assist the Superintendent in the carrying out of checking and sign-off of quality assurance Witness Points and Hold Points for the Works.

The Contractor shall co-operate with the CQA Engineer always, including the provision of access to all areas of the Works and at any locations, at the reasonable request of the Superintendent, for testing and verifications works, both on-Site and off-Site.

11.2. Contractor's Quality Assurance Representative

The Contractor shall appoint a full-time Quality Assurance Representative for these works, which will be that person's sole and full-time responsibility on the project.

The Contractor's Quality Assurance Representative shall be delegated sufficient authority, by the Contractor, to represent the Contractor in matters of Quality Assurance and Quality Control with respect to the Works, including the authority to direct the Contractor's works.

11.3. Contractor's Independent Testing Organisation

The Contractor shall engage an Independent Testing Firm to perform the tests specified to be undertaken in the Technical Specification. Technical Specification section 1.7.2 refers to this element.

The Contractor's Independent Testing Firm ("ITF") shall not be part of the Contractor's company. The Contractor shall procure a written statement of independence from their ITF stating that the testing firm is independent and has no financial interest in the Contractor, the Geosynthetic Lining Installer or any of the manufacturers/suppliers that are providing materials for the Works.

The ITF shall represent the Contractor for the performance of all verification testing required to be undertaken in accordance with the Contract including, but not necessarily limited to, the following:

- Quality control testing of materials at a material manufacturer's plant
- Quality control testing of materials following delivery to Site
- Manufacturer's documentation, including the procurement of certificates, test results and statements of conformance
- Field testing
- Field verification

The Contractor and the ITF shall work closely with the Superintendent's Construction Quality Assurance Engineer in relation to quality assurance, inspection and testing.

11.4. Work As-Executed (WAE) Drawings – Contract Drawings

The Specification, at section 1.13, details the requirements for the Contractor for the development and issue of As-Executed Drawings ("WAE Drawings") depicting the works.





The WAE Drawings shall be generated by the Contractor and shall include:

- a complete set of the electronic version of the Contract Drawings that depict the works as they are constructed. Additions, deletions and amendments shall be marked in red on these drawings, which are to be stored and distributed in Adobe 'PDF' format.
- All of the Contract Drawings shall be marked "Work As Executed", whether or not any additions, deletions or amendments have been included thereon.
- a complete set of all of the Contractor's drawings, surveys, and plans, developed during the progress of the Works, shall be updated and marked "Work As Executed"

The WAE Drawings are to be developed during the progress of the Works and will depict the details of the whole of the Works as the Works progress to their completion.

At the conclusion of each Stage and Separable Portion of the Works, the WAE Drawings shall depict the details of that stage as it has been created, constructed, cleared or remediated and in its final form.

11.5. Works As-Executed (WAE) Documents

Note that each area of the Site <u>that is excavated</u> by the Contractor, to remove materials, is to be represented by a computer-generated drawing containing dimensional data showing depth, surface footprint dimensions, and location data.

These computer-generated drawings shall include plans of remediated areas that have been cleared by excavation and reinstated to a landform matching the surrounding area, which includes -

- o the Capped Waste Stockpile area
- the Anode Pile area
- the Dickson Road stockpiles
- o East and West Surge Ponds
- other areas, cleared by excavation, that contain Miscellaneous Contaminated Materials

These computer-generated drawings shall be attached to a Ramboll Area Clearance Certificate.

A stage of the Works cannot qualify as being at a Stage of Practical Completion unless the WAE Drawings and/or computer-generated drawings for that stage and/or area have been completed, issued to and accepted by the Superintendent as being complete

12. Survey & Set Out of the Works

The Superintendent shall provide the Contractor with the information and survey marks necessary to enable the Contractor to set out the Works.

The Contractor shall carry out survey works in accordance with Section 1.11 of the Technical Specification, including detailed surveys for the setting out and the dimensional confirmation of the Works as the Works progress. Table 1-2 in the Technical Specification provides a detailed list of survey works required to be undertaken by the Contractor.

The Contractor shall maintain documentary records of all surveys and shall issue a copy of these records, at the time of their production, to the Superintendent for information.





13. Existing Services

The Contractor will be working across the Site where underground and overhead services exist, including on Dickson Road and the clay borrow pit.

To the Principal's current knowledge, most existing site services have been isolated from supply points during the decommissioning of the plant and the demolition works. However, the Contractor shall undertake their own investigations at each area of the Works to positively identify any existing services and their active state.

Services that will continue to be connected include:

- the three 132kV feeders that are connected to the Switchyard;
- the 415V supply to 12A pump house; and
- the temporary power supply feeding Building 21A and the surrounding buildings
- any other temporary services established by other Contractors on the Site.

Services within the vicinity of the ECC are limited to the high voltage (132kV Feeder 96W) overhead powerlines. These will remain live throughout the Works.

For the Contractor's excavation works an Excavation Permit is required, as part of the Hydro WHS Management Plan. Part of this permit process requires the identification of existing services and the Contractor shall ascertain specific services information relevant to that excavation.

14. Material and Resources to be supplied by the Principal

The Principal will make available, under certain terms, the following:

- Southern weighbridge as-is and available at the Contractor's discretion, to relocate on Site for load weighing and Gypsum dosing, and other load weighing requirements.
- Recycled drainage aggregate for use in the ECC including any Temporary Works or for the filling of voids around the Site as required.
- Certification of Environmental Clearance of areas of the Site that are remediated by the Contractor.
- Gypsum material, stockpiled at a location near the CWS, that is to be added to the CWS material.

15. Works Excluded

The following works are excluded from this scope of work:

- Transport, processing or handling of spent potlining and/or any other recyclable materials either within or from the Site for recycling;
- Demolition of remaining buildings and structures.
- Final environmental validation and certification for submission to the Site Auditor and for Project approval purposes

Schedule 8 Independent Engineer's Deed



Dated 21 Mary

2021

Independent Engineer's Deed

Minister for Planning and Public Spaces (ABN 20 770 707 468) Hydro Aluminium Kurri Kurri Pty Ltd (ACN 093 266 221) SMEC Australia Pty Ltd (ACN 065 475 149)

> Sydney . Melbourne . Brisbane . Cairns www.holdingredlich.com

H

Table of contents

1.	Definitions and interpretation	1			
	1.1 Definitions	1			
	1.2 Interpretation	4			
	1.3 Construction	5			
2.	Appointment of the Independent Engineer	. 5			
	2.1 Terms of appointment	5			
	2.2 Standard of performance of the IE Se				
	2.3 Representations and warranties	6			
	2.4 Fee and payment 2.5 Variations to IE Services	7			
	2.5 Variations to reservices 2.6 Defective IE Services	8 9			
3.	Term of appointment	9 10			
	3.1 Term 3.2 Termination	10 10			
-		10			
4.	Relationship	10			
	4.1 Nature of relationship	10			
	4.2 Independence	10			
	4.3 No conflicts of interest	11			
5.	Reliance				
6.	Certificate of Compliance	12			
7.	Representatives	12			
	7.1 Minister's Representative	12			
	7.2 Developer's Representative	12			
	7.3 Independent Engineer's Representation	/e 13			
8.	Suspension of IE Services	13			
9.	Indemnity	13			
10.	Incurance				
10.	Insurance	14			
	10.1 Public liability	14			
	10.2 Worker's compensation10.3 Professional indemnity insurance	15			
	10.4 Evidence of insurance	15 15			
11.	Dispute resolution	15			
	11.1 Not commence				
	11.2 Written notice of dispute	15 16			
	11.3 Attempt to resolve	16			
	11.4 Mediation	16			
	11.5 Court proceedings	16			
	11.6 Not use information	16			
	11.7 No prejudice	16			
	11.8 Continued Performance	16			
12.	Assignment and subcontracting				

Page ii

13.	Term	ination	17	
14.	Confi	Confidentiality		
	14.1	Confidential Information	17	
	14.2	Publicity	18	
15.	Notic	ces	18	
16.	Gene	ral provisions	19	
	16.1	Governing law and jurisdiction	19	
	16.2	Severance	19	
	16.3	Indemnities	19	
	16.4	Proportionate Liability	19	
	16.5	Waiver	19	
	16.6	Survival	19	
		Amendment	19	
	16.8	Entire agreement	19	
		Counterparts	20	
) Deed poll	20	
	16.11	L Limitation of liability	20	
Schee	dule 1	– Deed Particulars	24	
Schee	dule 2	– Land	26	
Schee	dule 3	– IE Services	27	
Schee	dule 4	– Certificate	29	
Schee	dule 5	– Fee Estimate Build-up	31	

This Independent Engineer's Deed is dated

21

2021

Parties:

Minister

Minister for Planning and Public Spaces (ABN 20 770 707 468) of 4 Parramatta Square, 12 Darcy Street, Parramatta NSW 2150

Developer

Hydro Aluminium Kurri Kurri Pty Ltd (ACN 093 266 221) of Hart Road, Loxford NSW 2327

Independent Engineer

SMEC Australia Pty Ltd (ACN 065 475 149) of Level 5, 20 Berry Street, North Sydney NSW 2060

Introduction:

- A The Developer owns the Land.
- **B** The Minister and the Developer have entered, or will enter, into the Voluntary Planning Agreement.
- **C** The Developer has engaged, or will engage, a Remediation Contractor to carry out the Remediation Works on the Land on the terms set out in the Site Remediation Contract.
- **D** The Voluntary Planning Agreement and the Site Remediation Contract contemplate that the Developer will appoint a construction quality assurance engineer to carry out services, equivalent to the IE Services.
- **E** The Independent Engineer has represented that it has the experience and expertise in the carrying out of the functions, services, duties and responsibilities equivalent to the IE Services.
- F The Independent Engineer has agreed to carry out the IE Services and its other obligations arising out of or in connection with this deed in accordance with the terms and conditions of this deed (Deed).

It is agreed:

1. Definitions and interpretation

1.1 Definitions

In this Deed, unless the context clearly indicates otherwise:

Access Road means the road to be constructed by the Developer providing access to the Containment Cell Land;

Address for Service means the address of each party appearing in Schedule 1 or any new address notified by any party to all other parties as its new Address for Service;

Approvals means all notices, orders, decisions, registrations, rulings, requirements, consents, permits, permissions, authorisations, certificates, licences, approvals and the like required from or given by an Authority or under any Legislative Requirement to carry out all or any part of the Remediation Works;

Authority means:

- (a) any governmental or semi-governmental or local government authority, administrative or judicial body or tribunal, department, commission, public authority, agency, Minister, statutory corporation or instrumentality; and
- (b) any other entity or person authorised by any of the above or by law to carry out or assist in the discharge of its functions in respect of Approvals;

Business Day means any day that is not a Saturday, Sunday, gazetted public holiday or bank holiday in Sydney, and concludes at 5 pm on that day;

Certificate of Compliance means the certificate (substantially in the form set out in Schedule 4 of this Deed) to be provided to the Minister and the Developer by the Independent Engineer, which, among other things, provides that the Remediation Works or relevant parts thereof:

- (a) have been constructed, installed and completed in accordance with the Project Documents, to the extent applicable to the Construction Remediation Works; and
- (b) comply with the applicable Legislative requirements;

Claim means any claim, demand, remedy, injury, damage, cost, loss, expense, liability, suit, action, proceeding, verdict, judgement, right of action or debt whether arising at law, in equity, under statue or otherwise;

Construction Remediation Works means so much of the Remediation Works as relates to:

- (a) the construction of the Containment Cell, including all design and temporary works, the construction of infrastructure associated with the Containment Cell (such as the installation of pipework for collection of groundwater, collection stumps for leachate and 'anchor' trenches) and the construction of the access road to the Containment Cell and the perimeter road, as described in State Significant Development Application SSD 6666; and
- (b) all works and activities referred to in Stage 1B;

Containment Cell means the containment cell to be constructed by or on behalf of the Developer on the Land in accordance with the Project Documents;

Containment Cell Land means that part of the Land to be transferred to the Minister, including the land on which the Containment Cell is located;

Corporations Act means the Corporations Act 2001 (Cth);

Developer is the party referred to in Item 2 of Schedule 1 of this Deed. A reference to the 'Landowner' in any Project Document is a reference to the Developer unless the context otherwise requires;

Development Consent has the same meaning as in the EPA Act;

EPA Act means the *Environmental Planning and Assessment Act* 1979 (NSW);

Fee means the fee calculated in accordance with clause 2.4;

Independent Engineer is the party referred to in Item 3 of Schedule 1 of this Deed;

IE Services means those services listed or referred to in Schedule 3 of this Deed and such other services as the Independent Engineer may be required to perform and provide under this Deed or the Project Documents;

Land means the land described in Schedule 2 of this Deed;

Legislative Requirements includes:

- (a) Acts, Ordinances, regulations, by-laws, orders, awards and proclamations of the jurisdiction where the Remediation Works or the particular part thereof is being carried out as amended or re-enacted from time to time including any such Act, Ordinance, regulations, by-laws, orders awards and proclamations made in replacement thereof;
- (b) certificates, licences, consents, permits, Approvals and requirements of organisations having jurisdiction in connection with the carrying out of the Remediation Works; and
- (c) fees and charges payable in connection with the foregoing;

Mediation Program means the Mediation Program of the Law Society of New South Wales as published on its website and as varied from time to time;

Minister is the party referred to in Item 1 of Schedule 1 of this Deed and includes the Minister's nominee;

Project means the development of the Land generally in accordance with State Significant Development Application SSD 6666, comprising (among other things) the remediation of the former Hydro Kurri Kurri Aluminium smelter site;

Project Documents means:

- (a) the Remediation Consent;
- (b) the Voluntary Planning Agreement;
- (c) the Site Remediation Contract and any other documents between the Developer and the Remediation Contractor in relation to the Remediation Works (including any scope of works, technical specification and remedial action plan); and
- (d) any other documents the Minister and the Developer acknowledge in writing to be a Project Document;

Remediation Contractor means the contractor appointed by the Developer under the Site Remediation Contract to perform the Remediation Works;

Remediation Works means the works as defined in the Site Remediation Contract and includes, without limitation, the Construction Remediation Works;

Remediation Consent means any Development Consent granted for the Project;

Site means the Land and any other lands and places made available to the Independent Engineer for the purpose of performing the IE Services;

Site Remediation Contract means the contract between the Developer and the Remediation Contractor for the performance of the Remediation Works;

Stage 1B Works means the following works and activities forming part of stage SP2 Part 1 of the Remediation Works:

- (a) creation of main East-West Haul Road;
- (b) creation of permanent creek crossing (culverts); and
- (c) construction of 5 dams 1 leachate dam, 3 sediment basins at containment cell site and 1 leachate dam at the capped waste stockpile;

Termination Event means:

- (a) a failure, refusal or inability of the Independent Engineer to perform all or any part of its obligations, duties and functions under this Deed, including within the time required under this Deed; or
- (b) a material breach by the Independent Engineer of any of its obligations, duties and functions under this Deed which:
 - (i) is not capable of remedy; or
 - (ii) is capable of remedy and the Independent Engineer fails promptly to rectify such breach within the period specified in a written notice issued jointly from the Minister and the Developer requesting the Independent Engineer to remedy the breach; and

Voluntary Planning Agreement means the agreement between the Minister and the Developer for the provision of development contributions by the Developer in connection with the Project.

1.2 Interpretation

In this Deed, unless the context clearly indicates otherwise:

- (a) a reference to **this Deed** or another document means this Deed or that other document and any document which varies, supplements, replaces, assigns or novates this Deed or that other document;
- (b) a reference to legislation or a legislative provision includes any statutory modification or substitution of that legislation or legislative provision and any subordinate legislation issued under that legislation or legislative provision;
- (c) a reference to the **introduction**, a **clause**, **schedule** or **annexure** is a reference to the introduction, a clause, a schedule or an annexure to or of this Deed;
- (d) **clause headings** and the **table of contents** are inserted for convenience only and do not form part of this Deed;
- (e) the introduction, schedules (if any) and annexures (if any) form part of this Deed;
- (f) **related** or **subsidiary** in respect of a corporation has the same meaning given to that term in the Corporations Act;
- (g) including and includes are not words of limitation;
- (h) the singular includes the plural and vice-versa; and

(i) words importing one gender include all other genders.

1.3 Construction

Neither this Deed nor any part of it is to be construed against a party on the basis that the party or its lawyers were responsible for its drafting.

2. Appointment of the Independent Engineer

2.1 Terms of appointment

- (a) The Developer appoints the Independent Engineer to perform the IE Services in accordance with this Deed.
- (b) Without limiting and in addition to the above, each of the Minister and the Developer appoints the Independent Engineer to perform the IE Services for the purposes of the Voluntary Planning Agreement.
- (c) The Independent Engineer's appointment under subclauses 2.1(a) and (b) commences on the date of this Deed and terminates on the date determined pursuant to subclause 3.2.
- (d) The Independent Engineer accepts such appointment under subclauses 2.1(a) and (b) and agrees to perform the IE Services in accordance with this Deed.

2.2 Standard of performance of the IE Services

In carrying out the IE Services the Independent Engineer must:

- (a) at all times act:
 - (i) in good faith, impartially, honestly and independently (without regard for the benefit of one party to the Deed over the other);
 - diligently and with a high degree of professional care, knowledge, experience and skill which may be reasonably expected of and in accordance with the standards applicable to a construction quality assurance engineer experienced in the performance of the same or similar services to the IE Services;
 - (iii) reasonably, fairly, professionally and honestly;
 - (iv) in a timely manner and with due expedition and without delay (and, in any event, within the times required under or contemplated by this Deed); and
 - (v) in accordance with this Deed and the Project Documents;
- (b) not waive or vary any requirements in the Project Documents;
- (c) not discharge or release a party from any of its obligations arising out of any Project Document;
- (d) comply with all Legislative Requirements;
- (e) exercise all reasonable skill, care and diligence expected of a properly qualified and competent professional rendering services of an equivalent nature to the IE Services and experienced in projects of a similar size, scope and complexity as the Project;

- (f) not infringe any intellectual property rights of the Minister, the Developer or any of their employees, officers, contractors, consultants or agents;
- (g) perform the IE Services in a manner that assists the Superintendent (as defined under the Site Remediation Contract) in carrying out its duties and functions, including in the carrying out of any checking and sign-off of quality assurance Witness Points and Hold Points for the Containment Cell, under the Site Remediation Contract;
- (h) without derogating from the Independent Engineer's obligations to the Minister under this Deed, comply with all reasonable directions and requirements of the Developer and its other contractors and consultants (including the Remediation Contractor) in relation to work health and/or safety (and/or, for the Developer only, in relation to the coordination and/or interface of activities) on or around the Site;
- (i) coordinate the performance of the IE Services with (and must not unreasonably interfere with, disrupt or impede) the activities being conducted or carried out by the Developer or its other contractors and consultants (including the Remediation Contractor) on or around the Site;
- not cause or contribute to, and will take all reasonable steps to prevent, any property loss or damage, and/or any personal injury, illness or death of or to any person, on or around the Site; and
- (k) promptly provide all information and documentation reasonably requested by the Minister and/or the Developer in relation to the performance of its obligations, or discharge of its liabilities, under or in connection with this Deed, including, by the last Business Day of each month during the performance of the IE Services a written report detailing:
 - (i) the Independent Engineer's actual progress in performing the IE Services compared to planned progress;
 - (ii) any known or foreseeable circumstances that may impact the Independent Engineer's performance of the IE Services, including the Independent Engineer's plans to mitigate any actual or potential impact of those circumstances; and
 - (iii) any other information that is materially relevant to the IE Services that the Developer or the Minister should reasonably be made aware of.

2.3 Representations and warranties

The Independent Engineer represents and warrants to the Minister and the Developer that:

- (a) it has the capability, expertise, skill, experience and ability to perform the IE Services, and that it will perform the IE Services in accordance with this Deed and otherwise comply with its obligations under this Deed;
- (b) it holds the appropriate qualifications, licences and registrations in accordance with the requirements of any Legislative Requirement necessary for the performance of the IE Services;
- (c) it has the power and has taken all corporate and other action required to enter into and authorise the execution of this Deed and the performance of the IE Services under this Deed;

- (d) it has not relied upon, and will not rely upon, any statement, information, documentation or representation made or provided by or on behalf of the Minister or the Developer on or prior to the date of this Deed;
- (e) no representation, warranty or information provided by it contains any untrue statement of material fact or omits to state a material fact necessary to make such representation and warranty not misleading in light of the circumstances under which it was made;
- this Deed constitutes a valid and legally binding obligation on it in accordance with its terms except to the extent of discretions of courts regarding the availability of equitable remedies and laws affecting creditors' rights generally;
- (g) the execution, delivery and performance of this Deed does not violate any existing law or any document or agreement to which it is a party or which is binding on it or any of its assets; and
- (h) it grants to the Minister and the Developer a perpetual, world-wide, irrevocable, royaltyfree and transferable (including a right to sublicence) licence to use, modify, copy and distribute any and all intellectual property rights in or relating to the IE Services and/or any information, deliverables and documentation prepared and/or provided by or on behalf of the Independent Engineer under or in connection with this Deed.

2.4 Fee and payment

- (a) The Fee will be calculated by the Developer in accordance with Schedule 5, based on the IE Services performed within the relevant period pursuant to this Deed.
- (b) The Independent Engineer may only issue payment claims for the payment of the Fee to the Developer in accordance with this subclause 2.4 and, in any event must ensure that every payment claim is issued:
 - (i) on the last day of each calendar month in which the IE Services are performed, for the Fee for the IE Services performed in accordance with this Deed within that month, priced in accordance with this Deed;
 - (ii) in the form of the payment claim prescribed in this Deed or, if none is prescribed, in the form reasonably required by the Developer, and must be accompanied with a tax invoice pursuant to all relevant Legislative Requirements;
 - (iii) in such detail as to enable the Developer to assess the IE Services performed, the period within which the IE Services were performed, and the calculations used (referencing the rates and prices set out in this Deed) in order to arrive at the Fee for the IE Services (including by identifying each separate disbursement comprised in the claim);
 - (iv) so as to not include any amount which was the subject of any previous payment claim or payment in connection with this Deed; and
 - (v) so as to not include any disbursements which were not approved in writing by the Developer prior to those disbursements being incurred or suffered by the Independent Engineer.
- (c) The Independent Engineer:
 - must promptly, after being requested by the Developer, provide all further information and documentation relating to any payment claim, and/or any IE Services the subject of any payment claim;
 - (ii) must perform the IE Services in accordance with the Fee Estimate Build Up (being the plan attached or referred to in Schedule 5, as updated by agreement in writing between the Developer and the Independent Engineer); and
 - (iii) despite anything to the contrary, and to the maximum extent permitted by law, acknowledges and agrees that:
 - (A) it will not be entitled to claim, or be paid for, any Fee or other payment for any IE Services performed outside of, or which exceed, the Fee Estimate Build Up (or any relevant item of the Fee Estimate Build Up), except to the extent that, as soon as possible and, in any event, prior to performing any such IE Services, the Independent Engineer notifies the Developer in writing of the need for the additional IE Services (or additional resources) and the Developer issues a written direction (pursuant to clause 2.5) agreeing to the performance of those IE Services (or additional resources);
 - (B) nothing in this Deed makes the Minister liable, or creates any liability for or in the Minister, to pay any amount, including anything connected with the Fee, arising from or in connection with this Deed and/or the IE Services; and
 - (C) the Developer is not required to pay for any IE Services and/or any Fee for any IE Services not yet performed, or disbursements not yet incurred by the Independent Engineer, as at the date of the payment claim, and/or any IE Services not performed in accordance with this Deed.
- (d) The Developer will assess each payment claim and, within 10 Business Days after receiving the relevant payment claim, issue a payment schedule to the Independent Engineer certifying its assessment of the amount due to the Independent Engineer (and, if that amount is different to the amount claimed, including reasons for this difference). The Developer will, subject to any right of set-off or deduction, pay the amount so assessed and certified as due, within 15 Business Days after receiving the relevant payment claim.
- (e) The Independent Engineer acknowledges and agrees that payment by the Developer is on account only and is not evidence of the value of work completed, an admission of liability nor evidence that the IE Services have been executed satisfactorily.
- (f) Despite anything to the contrary, and to the maximum extent permitted by law, the Developer may set-off or deduct, from any payment or amount due or payable to the Independent Engineer, any amount which is due or payable to the Developer under or in connection with this Deed.

2.5 Variations to IE Services

- (a) The Developer may vary, by adding or omitting, the IE Services at any time and from time to time in writing to the Independent Engineer.
- (b) The Independent Engineer must not vary the IE Services except to the extent necessary to comply with any written direction expressed as being provided under this subclause.

- (c) Where a variation to be directed under this subclause 2.5 reduces the scope, quality or quantity, or otherwise may adversely impact the quality of the IE Services and/or the Construction Remediation Works, the Developer must first obtain the written consent of the Minister for the relevant variation. The Minister must not unreasonably withhold or delay such consent.
- (d) If the Independent Engineer reasonably considers that a variation directed by the Developer under this subclause 2.5 will adversely impact on the quality of the IE Services and/or the Construction Remediation Works then, prior to it commencing any IE Services the subject of that direction, it must, as soon as possible, notify the Developer in writing and:
 - (i) the Independent Engineer must not commence any such IE Services until it receives written confirmation from the Developer to do so (if no such confirmation is given within 15 Business Days after the issue of the relevant notice by the Independent Engineer under this subclause 2.5(c), the relevant variation is deemed to have been withdrawn by the Developer); and
 - (ii) the Developer must obtain the written consent of the Minister for the relevant variation prior to issuing any confirmation under subclause 2.5(c)(i). The Minister must not unreasonably withhold or delay such consent.
- (e) Subject to subclause 2.5(c), the Developer will determine the value of the variation in accordance with Schedule 5 and the Fee will be adjusted accordingly.

2.6 Defective IE Services

- (a) The Developer may (and must where it is reasonably requested to do so by the Minister) direct the Independent Engineer to re-perform IE Services not performed in accordance with this Deed, and may specify a reasonable time for the Independent Engineer to complete the re-performance of the IE Services. The Independent Engineer must comply with such direction.
- (b) The Developer may give a direction under subclause 2.6(a) at any time up to and including the earlier of the date that:
 - (i) the Developer has issued the Final Certificate (as defined under the Site Remediation Contract) under the Site Remediation Contract; and
 - (ii) is 10 years after the date of this Deed,

or such longer period as agreed between the parties in writing.

(c) The Independent Engineer is not entitled to payment, including the payment of any Fee, for the re-performance of the IE Services in accordance with any direction given pursuant to this subclause 2.6.

3. Term of appointment

3.1 Term

The appointment of the Independent Engineer under subclauses 2.1 (a) and 2.1(b) of this Deed commences on the date of this Deed and, unless terminated earlier under subclause 3.2, continues until the later of:

- (a) the date Independent Engineer has completed and discharged all of its duties and functions under this Deed to the satisfaction of the Developer and the Minister; and
- (b) the earlier of:
 - (i) the Developer has issued the Final Certificate (as defined under the Site Remediation Contract) under the Site Remediation Contract; and
 - (ii) is 10 years after the date of this Deed.

3.2 Termination

This Deed will terminate immediately on the earlier of:

- (a) the termination of the Voluntary Planning Agreement (in which case this Deed terminates automatically); or
- (b) termination of this Deed in accordance with clause 13,

save in relation to any matter which remains outstanding and in progress as at the date of termination.

4. Relationship

4.1 Nature of relationship

- (a) The Independent Engineer is an independent contractor and is not an employee or agent of either the Minister or the Developer.
- (b) The Independent Engineer's employees, consultants and agents are not the employees, consultants or agents of the Minister or the Developer.
- (c) The Independent Engineer assumes full responsibility for the acts and omissions of each of its employees, officers, contractors, consultants and agents.
- (d) The Minister and the Developer are not liable for, nor will they be taken to have a liability for, or to have assumed or become (on enforcement of any of its powers or otherwise) liable for, the performance of any obligation of the Independent Engineer under this Deed or any associated document.

4.2 Independence

Without limiting or derogating from subclause 2.2(a), the parties agree that the Independent Engineer will act independently of all parties in connection with the performance of the IE Services.

4.3 No conflicts of interest

The Independent Engineer represents and warrants that:

- (a) it owes a duty of care and professional responsibility to each of the Minister and the Developer in connection with the performance of the IE Services;
- (b) it has disclosed to the Minister the contractual relationship it has with the Developer and, to the extent practicable, if a conflict of interest arises between the terms of this Deed and the separate agreement between the Developer and the Independent Engineer, the terms of this Deed will prevail;
- (c) subject to subclause 4.3(b) above, at the date of signing this Deed, no conflict of interest exists or is likely to arise in connection with the performance of the IE Services and the discharge of its duties and functions under this Deed; and
- (d) if, during the term of this Deed, any such conflict or risk of conflict of interest arises, the Independent Engineer will notify the Minister and the Developer immediately in writing of that conflict or risk of conflict and:
 - (i) take such steps as may be required by the Minister and the Developer to avoid or mitigate that conflict or risk; and
 - (ii) comply with all reasonable requests or directions of the Minister and the Developer in relation to such conflict or risk of conflict.

5. Reliance

The Independent Engineer acknowledges that the Minister and the Developer:

- (a) have entered into this Deed in reliance on, and are entitled to, and will, rely on:
 - (i) the independence of the Independent Engineer;
 - (ii) the particular skill, experience and ability of the Independent Engineer to perform all of the IE Services and otherwise comply with all obligations required by this Deed;
 - (iii) all deliverables prepared and provided by or on behalf of the Independent Engineer as part of the IE Services, including all reports and all Certificates of Compliance issued by it to the Minister and Developer in accordance with clause 6 of this Deed (acknowledging and agreeing that any and all deliverables may be provided to third parties on the condition that any reliance by the relevant third party on the deliverables will be at their own risk); and
- (b) may suffer loss if the Independent Engineer does not perform the IE Services in accordance with this Deed.

6. Certificate of Compliance

The Independent Engineer acknowledges and agrees that:

(a) as at the date of this Deed, the Remediation Works are divided into the following stages:

Stage No	Description
SP2 Part 1 (comprising Stage 1A and Stage 1B Works)	Construction of the Project infrastructure
SP2 Part 2	Construction of the containment cell stage 1 - base
SP2 Part 3	Performance of the site remediation and material transfer
SP2 Part 4	Construction of the containment cell stage 2 - capping, completion and compliance

- (b) in order to secure performance of the Remediation Works, the Developer has provided to the Minister security in the form of bank guarantees for each of the stages identified in subclause 6(a) above; and
- (c) it is a condition precedent to the release of the bank guarantee for stages SP2 Part 1, 2 and 4 that the Independent Engineer issue to the Minister and the Developer a duly executed Certificate of Compliance in respect of SP2 Part 1 (Stage 1B Works), SP2 Part 2 and SP2 Part 4.

7. Representatives

7.1 Minister's Representative

- (a) The Minister must appoint for the term of this Deed a Minister's Representative with authority to act on behalf of the Minister for the purposes of this Deed. At the date of this Deed, the Minister's Representative is specified in Item 4 of Schedule 1 of this Deed.
- (b) The Minister may replace or substitute the person appointed as the Minister's Representative by written notice to the other parties.

7.2 Developer's Representative

- (a) The Developer must appoint for the term of this Deed a Developer's Representative with authority to act on behalf of the Developer for the purposes of this Deed. At the date of this Deed, the Developer's Representative is specified in Item 5 of Schedule 1 of this Deed.
- (b) The Developer may replace or substitute the person appointed as the Developer's Representative by written notice to the other parties.

7.3 Independent Engineer's Representative

- (a) The Independent Engineer must appoint for the term of this Deed an Independent Engineer's Representative with authority to act on behalf of the Independent Engineer for the purposes of this Deed and the Project Documents. At the date of this Deed, the Independent Engineer's Representative is the person named in Item 6 of Schedule 1 of this Deed.
- (b) The Independent Engineer may replace or substitute the person appointed as the Independent Engineer's Representative by written notice to the other parties.

8. Suspension of IE Services

- (a) The Minister and the Developer may give the Independent Engineer a notice jointly instructing the Independent Engineer to suspend its performance of any or all of the IE Services under this Deed and the Project Documents until the Minister and the Developer give the Independent Engineer a notice jointly instructing the Independent Engineer to recommence performing the IE Services.
- (b) The Independent Engineer cannot make any Claim against the Minister or the Developer if any or all of the IE Services under the Deed and the Project Documents are suspended under this clause 8.

9. Indemnity

- (a) To the maximum extent permitted by law, the Independent Engineer agrees to, and must, indemnify, keep indemnified and hold harmless the Minister from and against all Claims that the Minister suffers, incurs, becomes liable for, or may suffer, incur or become liable for arising out of or in connection with:
 - loss or damage to property, or personal injury, illness or death arising out of or as a consequence of any act or omission of the Independent Engineer or its employees, officers, contractors or agents; or
 - (ii) the negligence or wilful default of the Independent Engineer or its employees, officers, contractors or agents.

The Independent Engineer's liability to indemnify the Minister will be reduced proportionally to the extent that any wrongful act or omission of the Minister, or the respective employees, officers, contractors or agents of the Minister, caused or contributed to the relevant Claim.

- (b) To the maximum extent permitted by law, the Independent Engineer agrees to, and must, indemnify, keep indemnified and hold harmless the Developer from and against all Claims that the Developer suffers, incurs, becomes liable for, or may suffer, incur or become liable for arising out of or in connection with:
 - loss or damage to property, or personal injury, illness or death arising out of or as a consequence of any act or omission of the Independent Engineer or its employees, officers, contractors or agents; or
 - (ii) the negligence or wilful default of the Independent Engineer or its employees, officers, contractors or agents.

The Independent Engineer's liability to indemnify the Developer will be reduced proportionally to the extent that any wrongful act or omission of the Developer, or the respective employees, officers, contractors or agents of the Developer, caused or contributed to the relevant Claim.

(c) This clause 9 shall survive the termination of this Deed and the completion of the Independent Engineer's duties and functions under this Deed.

10. Insurance

10.1 Public liability

- (a) On or before the date of this Deed or any earlier access to the Site for the purposes of performing the IE Services including any preliminary inspections of the Site (whichever happens earliest), the Independent Engineer must effect and maintain (or otherwise be insured under) a public liability insurance policy written on an occurrence basis with a limit of indemnity of the amount stated in Item 7 of Schedule 1 for each and every occurrence, which covers the:
 - Independent Engineer and its employees, officers, and agents (including for liability to each other); and
 - (ii) Minister and Developer for any vicarious liability that they may have for the acts or omissions of the Independent Engineer and its employees, officers, and agents,

for their respective liabilities for any:

- (iii) loss of, or damage to, or loss of use of, any tangible property (including property of the Minister or the Developer in the care, custody or control of the Independent Engineer for a sub-limit of the amount stated in Item 8 of Schedule 1 for each occurrence and in the aggregate for all occurrences in any 12 month policy period); and
- (iv) death of or bodily injury to any person including disease or illness to any person, (other than liability which is required by law to be insured under a workers compensation policy of insurance),

caused by, arising out of, or in connection with the performance of any obligation or the exercise of any right under this Deed.

- (b) The public liability policy must:
 - (i) be maintained until expiry of the defects liability period of the last separable potion to reach completion under the Site Remediation Contract; and
 - (ii) include cover for the insureds for liabilities assumed by them under the provisions of subclause 16.4 of this Deed.
- (c) For the avoidance of doubt, the limit of indemnity under a public liability insurance policy referred to in this subclause 10.1 may exceed the amounts stated in Item 7 or Item 8 of Schedule 1.

10.2 Worker's compensation

The Independent Engineer must effect and maintain workers' compensation insurance as required by law.

10.3 Professional indemnity insurance

- (a) Before commencing the carrying out of the IE Services, the Independent Engineer must effect and maintain (or otherwise be insured under) a professional indemnity insurance policy in a form and with an insurer acceptable to the Minister and the Developer with a limit of indemnity of the amount stated in Item 9 of Schedule 1 for any one claim and in the aggregate for all claims in any 12 month policy period.
- (b) The professional indemnity policy must:
 - (i) cover the liability of the Independent Engineer at general law arising from a breach of duty owed in a professional capacity, by reason of any act or omission of the Independent Engineer, its officers, employees, agents or subcontractors;
 - (ii) have a definition of professional services broad enough to include all professional services, activities and duties to be provided or performed by the Independent Engineer, its officers, employees, agents or subcontractors under this Deed; and
 - (iii) extend to cover claims for unintentional breaches of intellectual property rights;
 - (iv) extend to cover claims for unintentional breaches of consumer laws;
 - (v) have a retroactive date of no later than the earlier of the commencement of the work under this Deed or any preparatory work by the Independent Engineer, its officers, employees, agents or subcontractors; and
 - (vi) include cover for the insureds for liabilities assumed by them under the provisions of clause 16.4 of this Deed.
- (c) The Independent Engineer must maintain the insurance policy required to be effected under clause 10.3(a) from the date of this Deed until the date that is 7 years after completion of the IE Services under this Deed.

10.4 Evidence of insurance

The Independent Engineer must provide to the Minister and the Developer:

- (a) certificates of currency which contain sufficient detail to enable the Minister and the Developer to confirm that the insurances required to be effected by this clause 10 have been effected and are being maintained in accordance with this Deed; and
- (b) a letter from the Independent Engineer's insurer(s) confirming all of the Independent Engineer's insurance requirements under this Deed are and have been fully met and satisfied.

11. Dispute resolution

11.1 Not commence

A party must not commence any court proceedings relating to a dispute unless it complies with this clause11.

11.2 Written notice of dispute

A party claiming that a dispute has arisen under or in relation to this Deed must give written notice to the other parties specifying the nature of the dispute.

11.3 Attempt to resolve

On receipt of notice under clause 11.2, the parties must endeavour in good faith to resolve the dispute expeditiously using informal dispute resolution processes such as mediation, expert evaluation or other methods agreed by them.

11.4 Mediation

If the parties do not agree within 21 Business Days of receipt of notice under clause 11.2 (or any further period agreed in writing by them) as to:

- (a) the dispute resolution technique and procedures to be adopted;
- (b) the timetable for all steps in those procedures; or
- (c) the selection and compensation of the independent person required for such technique,

the parties must mediate the dispute in accordance with the Mediation Program. The parties must request the president of the Law Society of NSW or the president's nominee to select the mediator and determine the mediator's remuneration.

11.5 Court proceedings

If the dispute is not resolved within 60 Business Days after notice is given under clause 11.2 then any party which has complied with the provisions of this clause 11 may in writing terminate any dispute resolution process undertaken under this clause and may then commence court proceedings in relation to the dispute.

11.6 Not use information

The parties acknowledge the purpose of any exchange of information or documents or the making of any offer of settlement under this clause 11 is to attempt to settle the dispute. No party may use any information or documents obtained through any dispute resolution process undertaken under this clause 11 for any purpose other than in an attempt to settle the dispute.

11.7 No prejudice

This clause 11 does not prejudice the right of a party to institute court proceedings for urgent injunctive or declaratory relief in relation to any matter arising out of or relating to this Deed.

11.8 Continued Performance

Despite the existence of a dispute, the parties must continue to perform their obligations under this Deed.

12. Assignment and subcontracting

(a) The Independent Engineer must not assign this Deed or subcontract any part of the IE Services without the prior written consent of the Minister and the Developer, which may be given or withheld in their absolute discretion, and if given, on such terms as the relevant consenting parties think fit in their absolute discretion.

- (b) Notwithstanding any other clause of this Deed, the Independent Engineer shall be liable to the Minister and the Developer for the acts, defaults and omissions of the subcontractors, subconsultants and employees and agents of the subcontractors and subconsultants as if they were acts, defaults or omissions of the Independent Engineer.
- (c) The Developer may (and must where it is reasonably requested to do so by the Minister) direct the Independent Engineer to immediately cease using any person in the performance of any part of the IE Services if that person, in the reasonable opinion of the Developer (or the Minister):
 - has acted contrary to a law, Legislative Requirement or a procedure or policy of the Developer that the Independent Engineer was required to comply with under this Deed; and/or
 - (ii) has demonstrated a failure to perform his or her part of the IE Services to the standard of care required of the Independent Engineer under this Deed.
- (d) The Independent Engineer must promptly replace any person subject to a direction under this subclause with a person acceptable to the Developer (and, where relevant, the Minister) at no additional cost to Developer.

13. Termination

- (a) The Minister and the Developer may terminate this Deed:
 - (i) immediately by written notice to the Independent Engineer if a Termination Event occurs;
 - (ii) for convenience upon 15 Business Days' written notice to the Independent Engineer; or
 - (iii) if the Minister and the Developer reasonably consider that a conflict of interest has arisen or is perceived to have arisen in connection with the Independent Engineer in relation to the Project.
- (b) Termination of this Deed does not affect any accrued rights or remedies of any party.
- (c) If the Deed is terminated, the Independent Engineer and the Developer shall hold the Minister free and harmless against any liability and all claims in connection with the termination of this Deed.

14. Confidentiality

14.1 Confidential Information

- (a) The Independent Engineer must not disclose any proprietary or confidential information relating to the IE Services, the Voluntary Planning Agreement, the Site Remediation Contract, or the Project without the prior written consent of the Minister and the Developer, except as required by law.
- (b) This clause 14 shall survive the termination of this Deed and the completion of the Independent Engineer's IE Services and functions under this Deed.

14.2 Publicity

The Independent Engineer must:

- (a) not issue any information, document, advertisement or article in respect of the Project for publication in any media (including any form of social media);
- (b) not erect or make an advertisement on or in respect of the Project without the prior written consent of the Minister and the Developer; and
- (c) refer enquiries from the media concerning the Project to the Minister and the Developer.

15. Notices

- (a) Any notice, demand, consent, approval, request or other communication (**Notice**) to be given under this Deed must be in writing and must be given to the recipient at its Address for Service (including as applicable the Developer's Legal Advisor) by being:
 - (i) hand delivered; or
 - (ii) sent by facsimile transmission; or
 - (iii) sent by prepaid ordinary mail within Australia; or
 - (iv) in the case of a Notice to be given by the Developer or the Minister, sent by email.
- (b) A Notice is given if:
 - (i) hand delivered, on the date of delivery but if delivery occurs after 5pm New South Wales time or a day that is not a Business Day, is taken to be given on the next Business Day;
 - sent by facsimile and the sending party's facsimile machine reports that the facsimile has been successfully transmitted;
 - (A) before 5pm on a Business Day, on that day;
 - (B) after 5pm on a Business Day, on the next Business Day after it is sent; or
 - (C) on a day that is not a Business Day, on the next Business Day after it is sent; or
 - (iii) sent by prepaid ordinary mail within Australia, on the date that is 2 Business Days after the date of posting; or
 - (iv) sent by email:
 - (A) before 5pm on a Business Day, on that Day;
 - (B) after 5pm on a Business Day, on the next Business Day after it is sent; or
 - (C) on a day that it is not a Business Day, on the next Business Day after it is sent,

and the sender does not receive a delivery failure notice.

16. General provisions

16.1 Governing law and jurisdiction

- (a) The laws applicable in New South Wales govern this Deed.
- (b) The parties submit to the non-exclusive jurisdiction of the courts of New South Wales and any courts competent to hear appeals from those courts.

16.2 Severance

If any clause or part of any clause is in any way unenforceable, invalid or illegal, it is to be read down so as to be enforceable, valid and legal. If this is not possible, the clause (or where possible, the offending part) is to be severed from this Deed without affecting the enforceability, validity or legality of the remaining clauses (or parts of those clauses) which will continue in full force and effect.

16.3 Indemnities

- (a) Each indemnity in this Deed is a continuing obligation, separate and independent from the other obligations of the parties, and survives termination, completion or expiration of this Deed.
- (b) It is not necessary for a party to incur expense or to make any payment before enforcing a right of indemnity conferred by this Deed.

16.4 Proportionate Liability

- (a) The parties agree that Part 4 of the *Civil Liability Act* 2002 (NSW) does not apply.
- (b) The parties agree that their rights, obligations and liabilities will be those which would exist if Part 4 of the *Civil Liability Act* 2002 (NSW) did not apply.

16.5 Waiver

A right created by this Deed cannot be waived except in writing signed by the party entitled to that right. Delay by a party in exercising a right does not constitute a waiver of that right, nor will a waiver (either wholly or in part) by a party of a right operate as a subsequent waiver of the same right or of any other right of that party.

16.6 Survival

This subclause 16.6 and clauses 2, 4, 5, 9, 10.3, 14, 15, 16.1, 16.4, 16.5 and 16.10, and any other provision which ordinarily survives termination or repudiation of an agreement or deed, survive the termination or repudiation of this Deed.

16.7 Amendment

This Deed may only be varied or replaced in writing signed by each party.

16.8 Entire agreement

This Deed contains the entire agreement between the parties concerning its subject matter and supersedes all prior agreements and understandings between the parties in connection with that subject matter.

16.9 Counterparts

This Deed may be executed in any number of counterparts. All counterparts taken together constitute one instrument.

16.10 Deed poll

To the extent relevant, this Deed operates as a deed poll in favour of the Minister's nominee.

16.11 Limitation of liability

- (a) Subject to subclause 16.11 (b), the Independent Engineer's aggregate liability under this Deed to the Developer and the Minister is limited to AUD \$30,000,000.00.
- (b) Subclause 16.11(a) does not apply to any (and any calculation of liability for the purpose of subclause 16.11(a) will exclude all) liability:
 - arising from or as a consequence of any criminal, malicious or fraudulent act or omission of, wilful default of, reckless or intentional misconduct (done or omitted to be done having had regard to, or with conscious of reckless indifference to, the foreseeable harmful consequences arising from such misconduct) by, the Independent Engineer or any of its employees, officers, contractors, consultants or agents;
 - (ii) arising from or as a consequence of any breach of subclause 2.2(f) and/or 2.3(h);
 - (iii) arising from or as a consequence of the payment of any fine, penalty or charge, as issued or required by any Authority arising from or in connection with the IE Services, whether levied or imposed on the Minister, the Developer or the Independent Engineer;
 - (iv) arising from or as a consequence of any personal illness, injury or death and/or any loss or damage to property, including third party property (including personal or real property);
 - (v) arising from or as a consequence of the abandonment of the IE Services by the Independent Engineer;
 - (vi) for the Independent Engineer's internal and external legal costs;
 - (vii) for which insurance proceeds are recovered or recoverable under any policy of insurance required to be effected and/or maintained under this Deed, or would have been recovered or recoverable had the Independent Engineer complied with its obligations under this Deed with respect to effecting and maintaining certain insurance policies and/or had the Independent Engineer complied with its obligations under the insurance policies so effected and maintained (the Independent Engineer's liability of which will be limited to the relevant amount of insurance specified in Schedule 1); and/or
 - (viii) which by law cannot be contracted out of.

- (c) No party will be liable to any other party under or in connection with this Deed for any loss of business, loss of production, loss of profit or revenue, loss of contract or loss of goodwill, except for any such liability of the relevant party:
 - arising from or as a consequence of any criminal, malicious or fraudulent act or omission of, wilful default of, reckless or intentional misconduct (done or omitted to be done having had regard to, or with conscious of reckless indifference to, the foreseeable harmful consequences arising from such misconduct) by, the Independent Engineer or any of its employees, officers, contractors, consultants or agents;
 - (ii) arising from or as a consequence of any breach of subclause 2.2(f) and/or 2.3(h);
 - (iii) arising from or as a consequence of the payment of any fine, penalty or charge, as issued or required by any Authority arising from or in connection with the IE Services, whether levied or imposed on the Minister, the Developer or the Independent Engineer;
 - (iv) arising from or as a consequence of any personal illness, injury or death and/or any loss or damage to property, including third party property (including personal or real property);
 - (v) arising from or as a consequence of the abandonment of the IE Services by the Independent Engineer;
 - (vi) for which insurance proceeds are recovered or recoverable under any policy of insurance required to be effected and/or maintained under this Deed, or would have been recovered or recoverable had the Independent Engineer complied with its obligations under this Deed with respect to effecting and maintaining certain insurance policies and/or had the Independent Engineer complied with its obligations under the insurance policies so effected and maintained (the Independent Engineer's liability of which will be limited to the relevant amount of insurance specified in Schedule 1); and/or
 - (vii) which by law cannot be contracted out of.

Execution page

Executed as a deed

Signed, sealed and delivered by the Minister for Planning and Public Spaces (ABN 20 770 707 468), in the presence of:

Signature of witness

SAUSHYUN YANG

Name of witness in full

12 DARCY ST, PARRAMATTA USW 2150 Address of witness

Signature of the Minister for Planning and Public Spaces or delegate

NHITWO

Name of the Minister for Planning and Public Spaces or delegate

Signed, sealed and delivered by Hydro Aluminium Kurri Kurri Pty Ltd (ACN 093 266 221) in accordance with section 127 of the Corporations Act by being signed by those persons who are authorised to execute documents on behalf of the company:

.

..... Signature of Director

KICHARD BROWN

Name of Director

Signature of Director/Secretary

STEPHEN JAMES ROBERTS

Name of Director/Secretary

Signed, sealed and delivered by SMEC Australia Pty Ltd (ACN 065 475 149) in accordance with section 127 of the *Corporations Act* by being signed by those persons who are authorised to execute documents on behalf of the company:

Signature of Director

LLLIS Name of Director

Signature of Director/Secretary

Name of Director/Secretary

(2) 公司,并不可能把证明, (2) (2)

Page 24

Schedule 1 – Deed Particulars

Description	Details
Minister:	Department of Planning Industry and Environment
(subclause 1.1)	
Address for Service:	4 Parramatta Square, 12 Darcy Street, Parramatta NSW 2150
Email:	planningagreements@planning.nsw.gov.au
Contact	The Secretary
Developer:	Hydro Aluminium Kurri Kurri Pty Ltd
(subclause 1.1)	
Address for Service:	Hart Road, Loxford NSW 2326
Email:	richard.brown@hydro.com
Contact	The Directors
Developer's Legal Advisor	Gilbert + Tobin
(subclause 15(a))	
Address:	Level 35, Tower 2, International Towers Sydney
	200 Barangaroo Avenue, Barangaroo NSW 2000
Email:	info@gtlaw.com.au
Contact	Practice Group Leader, Real Estate + Projects Group
Independent Engineer:	SMEC Australia Pty Ltd
(subclause 1.1)	
Address for Services:	74 Hunter Street, Newcastle NSW 2300
Email:	newcastle@smec.com
Contact	Eric Wingate
Minister's Representative:	The Secretary
(subclause 7.1)	Department of Planning, Industry and Environment
Developer's Representative:	Richard Brown
(subclause 7.2)	
Independent Engineer's	Eric Wingate
	¢20,000,000
Fusic Liability insurance.	\$20,000,000
	Minister: (subclause 1.1) Address for Service: Email: Contact Developer: (subclause 1.1) Address for Service: Email: Contact Developer's Legal Advisor (subclause 15(a)) Address: Email: Contact Independent Engineer: (subclause 1.1) Email: Contact Minister's Representative: (subclause 7.1) Developer's Representative: (subclause 7.2)

.

H

Item	Description	Details
8.	The sublimit of public liability insurance for damage to property in the care, custody or control of the Independent Engineer must not be less than: (subclause 10.1(a)(iii))	\$20,000,000
9.	Professional Indemnity Insurance: (subclause 10.3)	\$20,000,000

Page 26

Schedule 2 – Land

Lot	Deposited Plan	Folio Identifier
Lot 3	456769	3/456769
16	1082775	16/1082775
318	755231	318/755231
319	755231	319/755231
411	755231	411/755231
412	755231	412/755231
413	755231	413/755231
414	755231	414/755231
420	755231	420/755231
769	755231	769/755231
1	456769	1/456769
2	456769	2/456769

1

Schedule 3 – IE Services

Without limiting the Independent Engineer's obligations under subclauses 2.2(a)(i) and 4.2, the Independent Engineer must discharge the functions, obligations, duties and services which this Deed and the Project Documents (including, without limitation, the scope of works and the technical specification which form part of the Site Remediation Contract and the CQA Plan) require to be, or contemplate will be, discharged by the Independent Engineer (or the 'CQA Engineer' or such similar terms used in the Project Documents), including, without limitation, the following:

- 1. Conduct inspections, tests and verifications of the Construction Remediation Works as is necessary to enable the Independent Engineer (or a highly skilled and experienced engineer performing services similar to the IE Services on a project similar to the Project) to be satisfied that the Construction Remediation Works have been constructed in accordance with the requirements of the Project Documents, including, without limitation, all inspections, tests and verifications to enable it to issue each Certificate of Compliance required under this Deed.
- 2. Observe, monitor, review and assess the quality of the Construction Remediation Works, including, without limitation, so as to allow and enable timely sign-off of witness and hold points within the Site Remediation Contract.
- 3. Perform the IE Services in accordance with the programme provided under the Site Remediation Contract, as updated from time to time by notice in writing from the Developer, so as to not unreasonably interfere with, disrupt or impede the Remediation Works.
- 4. Report to the Developer promptly after first becoming aware:
 - (a) any construction issues likely to have an adverse effect or affect on the quality of the Construction Remediation Works; and
 - (b) any act, matter or thing which has or is likely to have a material adverse effect or affect on the progress and provision of the IE Services (including the issue of the Certificate of Compliance required by clause 6 of this Deed and/or any of the deliverables under the CQA Plan), together with detailed particulars on how the Independent Engineer is dealing or proposes to deal with any such act, manner or thing,

and provide all information and documentation requested by the Developer in relation to the above.

- 5. At all relevant times independently verify that:
 - (a) the Containment Cell as constructed; and
 - (b) the quality of the Construction Remediation Works,

complies with the requirements of the Project Documents.

- 6. Exercise care, skill and expertise when carrying out independent quality checks, inspections, monitoring, reviews, tests, verifications and assessments on the Construction Remediation Works (including, without limitation, any elements and specific issues of the Construction Remediation Works) to satisfy the Independent Engineer that the quality of the Construction Remediation Works meets or exceeds the requirements of the Project Documents.
- 7. Monitor the construction activities on Site (to the extent they relate to, or are associated with, the Construction Remediation Works) and promptly identify, raise and document all observations to the Developer, which indicate non-compliance with the requirements of the Project Documents.

- 8. Satisfy itself that all activities in the process of the carrying out and completion of the Construction Remediation Works, including methods of work, sequence of activities, inspections and tests comply fully with the requirements of the Project Documents.
- 9. At the times required by this Deed, execute and issue a completed certificate of compliance in the terms and the form set out in Schedule 4 of this Deed.

Without limiting any of the above, the Independent Engineer must discharge all other services and tasks not described in this Deed and/or the Project Documents if those services and tasks are necessary for the carrying out of the IE Services in accordance with this Deed.

References to the 'CQA Plan' are references to the document titled 'Hydro Aluminium Kurri Kurri Pty Ltd Containment Cell Design Construction Quality Assurance (CQA Plan)' prepared by GHD dated October 2017, as incorporated into Appendix 3: Detailed Design Report of the Response to Submissions Report within the Project Documents.

Schedule 4 – Certificate

Certificate of Compliance

Issued as a prerequisite to the release of the Developer's bank guarantees

То:	Minister for Planning and Public Spaces (ABN 20 770 707 468) ("Minister")
	Hydro Aluminium Kurri Kurri Pty Ltd (ACN 093 266 221) ("Developer")
From:	SMEC Australia Pty Ltd (ACN 065 475 149) ("We", "our", "us" or "Independent Engineer")
Date of Certificate:	[#insert date]

The terms used in this Certificate (unless otherwise defined) have the same meaning as in the Deed (being the deed entered into between the Minister, the Developer and the Independent Engineer dated [insert date]).

We warrant and certify to the Minster and to the Developer that:

- (a) As at the date of this Certificate, we have carried out all periodic inspections and tests relevant to our obligations under the Deed and the Project Documents.
- (b) To the best of our knowledge and belief based on the periodic inspections and tests carried out to the date of this Certificate (and to what would reasonably be anticipated to be the best knowledge and belief of a highly skilled and experienced engineer based on the periodic inspections and tests which ought to have been carried out to the date of this Certificate under the Deed or the Project Documents), the Construction Remediation Works completed by the Remediation Contractor to the date of this Certificate:
 - reflect the intent and standards of quality of, and are constructed in accordance with, those elements set out in the Project Documents;
 - (ii) comply with the Project Documents and all Approvals (including those conditions of the Remediation Consent) which are applicable to the Construction Remediation Works;
 - (iii) are fit for any purpose and function expressly stated or implied in the Project Documents;
 - (iv) are suitable, appropriate, adequate and fit for the purposes of the Project as stated in the Project Documents;
 - (v) comply with all applicable Legislative Requirements and Australian codes and standards as well as the requirements of all relevant Authorities; and
 - (vi) do not include any changes, except as a result of variations, as noted in subparagraph
 (c) below.

(c) The Construction Remediation Works periodically inspected and/or tested by us to the date of this Certificate, substantially comply, to the best of our knowledge and belief based on the periodic inspections and tests carried out to the date of this Certificate (and to what would reasonably be anticipated to be the best knowledge and belief of a highly skilled and experienced engineer based on the periodic inspections and tests which ought to have been carried out to the date of this Certificate under the Deed or the Project Documents), except as noted below, with the requirements of the Project Documents:

[list any variations, departures, defects or non-conformances].

- (d) To the best of our knowledge and belief based on the periodic inspections and tests carried out to the date of this Certificate (and to what would reasonably be anticipated to be the best knowledge and belief of a highly skilled and experienced engineer based on the periodic inspections and tests which ought to have been carried out to the date of this Certificate under the Deed or the Project Documents), the Remediation Contractor has (where relevant):
 - (i) complied with and satisfied the requirements in the Project Documents in respect of the quality of the Containment Cell; and
 - (ii) constructed the Containment Cell in accordance with the Project Documents.
- (e) The IE Services, as performed by us to the date of this Certificate, have been completed to the extent required by the Deed and the Project Documents, and comply with the Deed and the Project Documents.

A reference in this certificate to periodic inspections and tests includes a reference to all tests, verifications, assessments, quality checks, and monitoring referred to in Schedule 3 of the Deed

We acknowledge and agree that this Certificate is being provided for the benefit of the Minister and the Developer, jointly and severally, and that the Minister and the Developer are relying and will rely on this Certificate.

This Certificate is without limitation to, and is being provided in addition to, the Deed and does not in any way prejudice or limit the Minister's and/or the Developer's rights and entitlements under or in connection with the Deed.

Signed, sealed and delivered as a **Deed Poll** by SMEC Australia Pty Ltd (ACN 065 475 149) in accordance with section 127 of the *Corporations Act* by being signed by those persons who are authorised to execute documents on behalf of the company:

.....

Signature of Director

Signature of Director/Secretary

.....

Name of Director

Name of Director/Secretary

S:10980593_1 NRA

Schedule 5 – Fee Estimate Build-up

S:10980593_1 NRA

Schedule 5

This Schedule 5 is the Fee Estimate Build Up for the purposes of clauses 2.4(c)(ii) and (iii), and this Schedule 5 is to be read subject to clauses 2.4(c)(ii) and (iii). For the avoidance of doubt, and without limiting the preceding sentence, any disbursements incurred by or on behalf of the Independent Engineer will at all times be subject to clause 2.4(c)(iii)(C).

To the extent there is any discrepancy, ambiguity or inconsistency in or between this Schedule 5 and any other document forming part of this Deed, the highest standard and/or most onerous obligation will apply to the Independent Engineer.

References to "Consultant" or "SMEC" (or similar expressions) in this Schedule 5 are deemed to be references to the Independent Engineer.

References to "Services" in this Schedule 5 are deemed to be references to the IE Services.

References to "Contract" in this Schedule 5 are deemed to be references to this Deed.

References to "Hydro" in this Schedule 5 are deemed to be references to the Developer.

5.1 Pricing

5.1.1 Target Cost

The target costs below represent the Consultants best and most accurate estimate of the total cost to deliver the Services and:

• Excludes GST and is stated in Australian currency (\$AUD);

• Is on a time and disbursement basis and shall not be exceeded except by way of an approved variation, provided in accordance with the Contract;

• Includes all subcontractor and subconsultant hours and costs if applicable;

• Is all-inclusive of the costs and disbursements associated with the Consultants satisfaction of all of its obligations arising under the Contract; and

• Is not subject to rise and fall (unless the Contract otherwise expressly requires).

SUMMARY OF TARGET COST Item	Work Description	Cost (\$AUD)	Estimated program Period of I.E./CQA involvement
1.1.1	Project Management	\$107,320	89 weeks
1.1.2	SP2 P1 – Construction of the Project Infrastructure	\$10,230	3 weeks
1.1.3	SP2 P2 – Construction of the Containment Cell Stage 1 – base	\$188,370	33 weeks
1.1.4	SP2 P3 – Performance of Site remediation and material transfer	\$35,210	29 weeks
1.1.5	SP2 P4 – Construction of the Containment Cell Stage 2 – capping, completion and compliance	\$105,170	24 weeks
1.16	Vehicle Allowance. SMEC has assumed a vehicle allowance of \$1,100 per month for a 24 month	\$26,400	24-month total period

	period (approximately the total construction period). This will be charged as a separate cost on a monthly basis		
TOTAL		\$472,700	89 weeks

Notes:

• The fee estimate is based on 2020/2021 rates.

• The fee estimate was developed in accordance with the Draft Construction program provided in Addendum 1, and SMEC estimated the average input over the period that the I.E./CQA will be involved is as follows:

• CQA Engineer (PM) 11.7 hrs/wk

o CQA monitor 24.0 hrs/wk

o Senior Engineer 2.4 hrs/wk

o PD / Technical 2.4 hrs/wk

A detailed breakdown of the fee estimate is as follows:

27031 17 00 27031 17 00 27031 17 00 27031 17 00 27031 17 00 27031 17 00 27031 17 00 27031 17 00 27031 17 00 270410 8 Cost (\$AUD) sclusive of GS Ho days I cay D cays Cays D cays D cays 2 Strange 0. Discontinue Approximation SEPARABLE PORTION 2 PART 2 - CONTAINMENT CELL STAGE 1 containent citu, 1 2. Let Dirac recipe (num transfer part) for a recipe (num transfer part) for a recipe (num transfer part) (num transfer par Procure Temporary Water Treatment Plant (TWTP) SEPARABLE PORTION 2 PART 1A SEPARABLE PORTION 2 PART 18 Charter and Grubbing Installation of Saura Exclusion Fercing Installation of Sediment Fercing & Losion Lesvation of Contaliment Cell Cutto fill Esthwork Contament Ce **(EY DATES & MILESTONES** di S00mm x 500mm and g Subcontractor Mobile z 1 - Province of 100mr Swale Drain including Install Jute Matting PROJECT WIDE ITEMS Hydro issue Daracon with Letter to Lavate Swale Drain includie Licavate Swale Drain includie Liupply & Install Jute Matting UMMARY of TARGET COS level of attend-ar 8 1 54 -----1/4 1/4 1/4 List-all'investigation of concertation meet CLAP income PM for separable portion - Higher PM allibuarce for PM for separable portion - Higher PM allibuarce for PM for separable portion - Converge PM during 43 PM for separable portion - Average PM during 43 PM for separable portion - Average PM during 14 PM for separable portion - Average PM during 14 liaton with Superintensent and Con MQA, other submittals, with essing o sample and costent (cotte) to lab treating (coste) review & approval, in-other support periodic inspections (scalable) periodic inspections (scalable) periodic inspections (scalable) h shows a 9 hr/day incl breaks). Actual time to be claimed. Aruction program " provided by Hydro. CHECK below: andum 1]. NBI, Daracon program has start-to-finish errors Final report only entire D UNMENT CELL STAGE 1 IMEDIATION & MATCRIAL TRANSFER UNMENT CELL STAGE 2 & COMPLETION start-up allowance MQA review for entire project Assume all sampling in 1 mobil 1 inspection included above excluded above included above i diw quire Inspection Inspection urther detail 2020/21 rate: cl in PM inspection = = ~ 3 <u>8</u> 8 250 220 PD / Technica 2 BE 5 level of attendance allowed.for: a hrs/day, 5 day week (per Daracon to approx. weeks - estimate from "pr or "Timeframe from Daracon updated 110 \$ 2111 24.0 1 meets 1 m COA monito Art 5/W (1039/21) 5 13 5 Nath Part 100 ALL M (111 - 100 ALL M (111 - 100 - 111 8 14 164 164 120 CQA Engineer (PM) RATE S/hr (2020/21) \$ Alte work profile program update program program (1 workwork) program (1 workwork) program (1 workwork) with CoAlmer transform PM 5/wk 5 35,495 1,010 1,010 3,920 2,700 1,480 35,840 21,340 21,340 26,160 1,040 6,640 1,560 6,640 1,010 1,010 1,010 7,200 07,320 • 5 (plus vehicle) \$ (plus vehicle) \$ 3 11 5 prog days where used in fee calc (plus vehicle) [plus vehick 11111

L

6855

Operation Operation <t< th=""><th>Control (Control (Contro) (Control (Contro) (Control (Contro) (Contro) (Contro) (Contro)</th><th></th><th></th><th></th><th></th><th>ACCANALING ACCANALING ACCANAL</th></t<>	Control (Control (Contro) (Control (Contro) (Control (Contro) (Contro) (Contro) (Contro)					ACCANALING ACCANAL
1 649 2 6491 2 6491 6 6491 1 6491 1 649 1 649 1 649 1 649	1 6 10 1 6 10 1 6 10 5 6 10 5 6 10 1 6 10 2 6 10 1 7 10	2 0 mm 2 0 mm 2 0 mm 2 0 mm 2 0 mm 4 0 mm	16 carr 1 car 1 car 1 car 1 car 1 car 1 car 1 car 1 car 2 cars 2 car 2 car 2 car 2 car 2 car 2 car 2 car 2 car 2 c	36 cansi 26	15 a.m. 15 a.m. 16 a.m. 16 a.m. 16 a.m. 16 a.m. 16 a.m. 5 a.m. 5 a.m. 2 a.m. 2 a.m. 2 a.m. 2 a.m.	1000 1000 1000 1000 1000 1000 1000 100
Control 10 and a second dependence on convergence and approximate processing and a second	The distribution of participant equation and a second second equation and a second second equation and a second se	 Tori I. Format Elimination State States strates strates strates and strates strates strates strates and strates s	Activity of the left part of data latter the degree of the left part of data latter to be considered in the latter of data latter to be considered in the latter of the latter of the latter to be considered in the latter of the latter of the latter to be considered in the latter of the latter of the latter of the latter to be considered in the latter of the l	Class description fraction description and description	Concerning and the set of a direct dataset. The set of direct dataset	 Christian Statikan Statika
55 555 55555						555555555555555555555555555555555555555
The server of the class asserted that system / human proce trapection (scatalor) for share or stands proce trapection (scatalor) for share or stands process asserted to classific the stands of stands process asserted to classific the stands of stands process asserted to classific the stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of process asserted to classific the stands of stands of stands of stands of process of stands of stan	Induction International International International International International International International International International International International International International International International International International	Примого в сполодитиристири на при измер Примого в сполодитиристири на при измер при измер на сполодитири на при измер при измер на при измер на при измер на при измер при измер на при измер на при измер на при измер при измер на при измер на при измер на при измер при измер на при измер	province in a province of a pr	 инструментации советствание полном учите советствание по	A representation of the second	An enter performance interactional and an enter performance in a state of the state
таката пред станција на пред станција н	Access the of counter states and the other of counter states and the statestates and the statestatestatestatest	 Colori abovi Color	Inclusion above to concentrational content concentration to concentrational concentration to concentrational concentration to concentrational concentration to concentrational concentration and concentrational concentration and concentrational concentration and concentrational concentration and concentrational datasets in cuto datasets in cuto	Полну почита на селото на селот Полну почита на селото на селото селото на селото на с селото на селото на сел На селото на селото на На селото на селото н	то по може по се	O) Insert every transmission (LL), differences and transmission (LL), differences a
1.00 6 M	7.786 69 244 145 11 11 11 11 11 11 11 11 11 11 11 11 11		800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800 - 800			
55	2	v				

HUMONI I M UUNONI I M LUMONI I M UUNONI I M <t< th=""><th></th><th>ANALOGE IN CONTRACTION ANALOGE IN CONTR</th><th></th><th></th><th>2002/12/02/12/02/22/22/22/22/22/22/22/22/22/22/22/22</th><th>A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1740 A MONTANTING 1890/</th><th>10/06/2011/202/001 10/06/2011/200 27106/20211/200 27106/20211/200 27106/102111/200 27106/12111/200</th><th></th><th>22262 0011100/J0015 00711205/J0015 00711205/J0015 22852 00111202/J0011 22852 00111202/2011</th><th></th></t<>		ANALOGE IN CONTRACTION ANALOGE IN CONTR			2002/12/02/12/02/22/22/22/22/22/22/22/22/22/22/22/22	A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1730 A MONTANTING 1890/AR1740 A MONTANTING 1890/	10/06/2011/202/001 10/06/2011/200 27106/20211/200 27106/20211/200 27106/102111/200 27106/12111/200		22262 0011100/J0015 00711205/J0015 00711205/J0015 22852 00111202/J0011 22852 00111202/2011	
		Trapport Verter Transmission Example Example <thexample< th=""> Example <thexam< td=""><td></td><td>titater Ins tookspin GA. Inset Time on de Sockspin in GC Facilitier - Sai</td><td></td><td></td><td></td><td>CVS the locary task, but does contract and non-function in CVS (national CVS) and another construction of the CVS (national ACC) and another construction of the CVS (national ACC) and another construction of the CVS (national ACC) and another construction and another the ACC) and another and another another acc) and another another another acc) and another another acc) and another another another acc) and another another another acc) and another another another acc) and another another another acc) and another another acc) and another another acc) and another another acc) and another another acc) and another acc) and another acc) and another acc) and another acc) and another acc) and another acc) another acc) anoth</td><td></td><td>are entrusis contras. Las bio la vertila sual toma Succipite to Landid Roopent</td></thexam<></thexample<>		titater Ins tookspin GA. Inset Time on de Sockspin in GC Facilitier - Sai				CVS the locary task, but does contract and non-function in CVS (national CVS) and another construction of the CVS (national ACC) and another construction of the CVS (national ACC) and another construction of the CVS (national ACC) and another construction and another the ACC) and another and another another acc) and another another another acc) and another another acc) and another another another acc) and another another another acc) and another another another acc) and another another another acc) and another another acc) and another another acc) and another another acc) and another another acc) and another acc) and another acc) and another acc) and another acc) and another acc) and another acc) another acc) anoth		are entrusis contras. Las bio la vertila sual toma Succipite to Landid Roopent
Alter and a direct control in a direct of the direct of th	ave Second Second Second Second Second	In the second se	And the first state of the second state of the first state of the second state of the	per CGAP. Processing scalable	equators or anglement approx (QAP Neuraliy cubic) (1) equators or agriculture to QAP Neuraliy schedule (1) 1) even of the most cubics of the cubic cubics (1) 1) even of the most cubics of the cubic cubics (1) 1) even of the most cubics of the cubic cubics (1) 1) even of the most cubics of the cubic cubics (1) 1) even of the most cubic cubics (1) 1)	Construction of the state of th	An and a control to Addition Additional addition	inspection municipalization to pre COMP Palanticle under	majanstans nu neg plusement as per Caule Palanstally u stabile Prin majanstans nu neg plusement as per Caule Palanstally u stabile Pri	And a start of the start is per CLAP Polaritaly usable
All service and all a prive allocate interactional product allocate and allocate an	Per previous conception priority in 2004 No. 100 MIGRIS WITH FRAME DRAFT Prior Priority Conception and Concepti	too and to the second se	policited above from accurate accurate	inducer above too second account	indicate above for exercise extensional between above for exercisent extensional between the exercisent extension to the extension above for exercisent extension	Leaners formous of needs were sub-	TIC Includes Joben (to concurrent activitien) TIC Includes Joben (to concurrent activitien)	denar valating these sales	dinardy nagacting them cales dinargy nagacting them cales	ROME THE RECOMPTION OF THE
1 100 100 100 100 100 100 100 100 100 1										

Π

22/02/2021 17:00 30525		12/05/202117.00 312		21/05/2021 17 00 314	24/05/2021 17 00 316							18/05/2021 17:00 32455	1/07/2021 17:00			1/07/2021 17.00 329	29/01/2021 17:00	12/02/2021 17:00			202 00.11.1202/2016 M11 00211202/2021		12/02/2021 17:00 337FF		23/07/2021 17:00	00.11 1707/10//	29/01/2021 12:00 340FF	EPE 00/21 12/02/20/21	20/04/2021 12/02 142524 00:21 12/02/20161	20/04/2021 17:00 34655	20/04/202117.00 3465510	11/02/2021 17:00 349FF 5 Gav	5055+10 eu	5/07/2021 17:00 351			19/02/2021 17:00				100/11/202/20/00 8/02/202117:00	2/02/2021 17:00 340	4/02/2021 17:00 363				16/07/2021 17:00 368		23/07/2021 17:00 371	14/07/2021 17:00	29/01/2021 17:00	5/07/2021 17:00	5/07/2021 17:00 35245	22/02/2021 17:00 37555	22/02/2021 17:00 379FF	22/02/2021 17:00 375;380;379	186 00.71 1202/20/22		2/02/2021 17:00 375	286 00.71 1202/20/4	****		
7/05/2021 8:00	11/05/2021 8:00	12/05/2021 8.00	13/05/2021 8.00	17/05/2021 8.00	24/05/2021 8.00	25/05/2021 8 00	00.81202/20/02	16/04/2021 8 00	3/05/2021 8.00	4/05/2021 8.00	18/05/2021 8.00	18/05/2021 8:00	00:8 1202/50/52	25/05/2021 8:00	23/06/2021 8.00	30/06/2021 5.00	00-8 1202/10/82	28/01/2021 6.00	28/01/2021 8.00	4/02/2021 8.00	8/02/2021 8:00	25/01/2021 8:00	1/02/2021 8.00		25/01/2021 4:00	00:8 1707/10/67	25/01/2021 8 00	1/02/2021 8:00	18/02/2021 8:00	18/02/2021 8:00	18/02/2021 8.00	14/04/2021 8.00	28/04/2021 8:00	28/06/2021 8.00	15/02/2021 8.00	15/02/2021 8:00	16/02/2021 8:00	8/03/2021 8:00	10/03/2021 8.00	28/06/2021 8:00	1/02/2021 8:00	1/02/2021 \$ 00	002/2021 8:00	8/07/2021 8.00	R/07/2021 S-00	13/07/2021 8-00	16/07/2021 8.00	19/07/2021 8:00	23/07/2021 8:00	00.8 1202/10/52	25/01/2021 8.00	25/01/2021 8:00	7/06/2021 8.00	25/01/2021 8:00	26/01/2021 8.00	22/02/2021 17:00	22/02/2021 17:00	1/02/2021 8:00	1/02/2021 8:00	00.8 1202/2016	2/02/2021 8.00	one a souther the	
2 days 13 days	Ved 1	2 GAV	2 days	SAED S	1 day	Van 1	23 days	11 days	1 Cav	10 cays	1 cay	Ven I	20 Gays	20 Gavs	skep s	2 caps	Circle C	12 days	skep s	Aro 1	step 5	14 days	10 cays		124 Gays	SAPD YTT	5 cays	10 days	40 days	40 cays	10 days	20 days	42 caps	5 Cave	24 Cays	1 Cav	1 Cays	S Gays	7 Cays	3 days	e days	2 days	2 GAVE	12 days	3 days	2 cays	1 Cay	4 caps	1 cay	117 6475	skep s	the early	10000	20 days	20 days	SAED O	0 Gays	26 days	S days	2 Gave	7 Gays		
Hast imaterial, place are compact at ECC in controllen manner ARIA CAST OF CLAY BORNOW PTT (ACCOD)	Remove vegetation	restore the support of the second state of successing as successing and the second second state of the support	Haul, place and compact at ECC in controlled manner	Environmental Clearance Validation and procure certificate for AEC30	bite Recovery - Ioad, Paul, place & compact clean material from Stockpile to AFC10 Footprint Annh native ersissisant mer area	water between a set of the set of	EAST SURGE POND AND DRAINAGE LINE (ACCS AND ACC B)	Construct a dewaterling facility, including drainage to Temporary leachate Storage Basin	Excevete 200mm sediment to Temporary Dewatering Area	Allow Sediment Material to Dewater	bewatered	Haul Gewatered material, place & compact at ECC in controlled marrer MISCELLANTONIS CONTAUINATED MATERIALS	Verity all area's reduring cleararce, excavate to 500mm repth, load to ECC	Haul, place and compact att ECC in a controlled manner	Environmental Clearance Validation and procure certificate for all areas	bite Recovery - load, houl, place & compact clear material from Stockpile to contaminated area footprints. Readvand Arcenterior Cleanance	Clear the surface of roads and accessways by approved methods	Gas Management System - Part 1 of 2	install Moreontal Gas Trench within top of waste material, incl. DN160 slotten page surrounden by recycled aggregate Jost of Detrois non-tion and with a survey and an anomalication	instant exterior was a connect states (April 2) and the conne	Install Vertical Gas Bores at points on each radial Horizontal Gas Trench - DN 300 bore through waste material	QA Decumentation	Prepare & Finalise QA and SAS Documentation QA and SAS Documentation Approval	SEPARABLE PORTION 2 PART 4 - CONTAINMENT CELL STAGE 2 & COMPLETION	Pleement of ECC Cassing Laws	22 days from "retworkdays (X,X)" LINE INSERTED	Lirer Subcontractor Mobility to Site	Laber 1 - Install Geonet Disurage Composite Laver 2 - Plant are commont Thinma Calmential Survival Lawer	Layer 3 - Install Geosynthetic clay liner	Layer 4 - Install LLDPC geomembrane	taver 5 - ritoretison geotreatur Laver 6 - Althum Recyclen niainage aggregate, from MAKK onsite stocknike, haul ann niace	Layer 7 - Separation geotextile	Layer 8 - Load from stockpile, place and compact 1.3m thick subsoil layer	Läver 9 - import, load and place 100mm topsoil Avenomich Topsoil	Gas Management System - Part 2 of 2	Install DM150 PVC pipe from Distribution Box to Antating Enect Cowel	instant notating i cuci coveri install 1500 × 1500 × 1500 mais concrete footing	Install fem galvanssen Steel nuctif. Enuct Cowel	Install Gas Montourg bores	Install a DN140 SMI Api pipe along the invert of the CCC Perimeter Dianase Charcel through Galeert 036.01	Decommissioning & Removal of the Leachate Duffer Storage Dam	Decommission pipework, advivery system to the Leachate Dum Remove HDPT Liner to Call Prior to completion of ECC Capperg Remove Vehicle Turnationnel Part D. and transfer missionil to Constant District Constants.	toad from stockpie/Paul, pisce& compact till to teachate Buffer Storage Dam voir & revbate area	Decommissioning & Removal of Sectiment Basin 1, 2 & J	Load and hauf from stockpile, place and compact fill to Sediment Basin 1 void and reshape area	core and nauritiom socieptie, place and compact till to Sectiment Ravin 2 void and reshape area. Provide surface contour for water flow from Act one autier Cubiert 2 to Swith 2 arrows Sectiment Barin 1 Contenue	Provide surface contour for water flow from Agi pipe outlet Culvert 3 to Swale 1, across Sectiment Rasin 2 Footprint	Load and hauf from stockpile, place and compact fill to Seciment Basin's voiri and reshape area	Provide surface contour for water flow from Swales 03 and 04 to the existing water course	4A Decumentation Preparek Finalise DA Decumentation for Handover	QA Documentation for Handover Approval	Preliminaries and Administration. Firsting is protocome Management Plan & Work Mathew Streament	research and the present restore a voir whethout statement. Decommissioning Work Method Statement	Completion of the Project Data Report	Provide all Manufacturers' and Suppliers' Warranties and Guarantres	Practical Completion		Removal of Temporary Works	Removal and disposal, off-Site, of all Temporary fercing Clarance of Temporary Storbulls areas, including disposance of an include to very vision the Site of Jerussian	Removal of Temporary Accessways and the reinstatement of the surface to match the surfacement	Removal of Temporary Stortwater Runoff and Sediment Controls		Downed: All Contractions along and action Trans Boot Confidence and Other Trans Information 1, and a second second
U.A.			Lid									14		1/4					13	F/T	1/1		1/4					Lid	113	5	1/4	14	1/4	1/4		Lid	L/d	L/d	1/d	P/T		Lid								1/1					La								
etters Australia - John Lad or Launaved Saran antivadon			inspection during placement as per COAP. Potentially scalable									ADDED AND A AND THE REAL PROVIDED AND ADDED AND ADDED ADDED		inspection ouring placement as per CQAP. Potentially scalable					periodic inspections (scalable) periodic inspections (scalable)	periodic inspections (scalable)	periodic inspections (scalable)	in the second of all states accounting the second states of a state of the second states of the second states accounting the second states are second states as a second state of the second states are second s	receipt of all cata, progressive compliation of validation report, IR signal			variable. Includes items highlighted yellow below	IT measures for critical commonweaks of any figure function	FI prevencer for critical components of gas / cap / barrier periodic inspections (icalable)	FT presence for critical components of gas / cap / barrier	FT presence for critical components of gas / cap / barrier ET ensuence for critical commonants of any / cap / burrier	periodic inspections (scalable)	FT prevence for critical components of gas / cap / barrier	periodit inspections (scalable) periodic inspections (scalable)	periodic inspections (scalable)		periodic inspections (scalable) periodic essertions (scalable)	periodic inspections (scalable)	periodic inspections (scalable)	periodic inspections [scalable]	perioric expections (scalable)		periodic inspections (scalable)								receipt of all data, progressive compilation of validation report, if supoff					allowance for review & comment								
We want the state of the state			already inspecting these dates								alteady inconcline these sufes			already inspecting these dates					1 inspection	1 inspection	1 inspection	4 overlap with insections. Final report only				F/T presence during this period (CQA monitor). CQA & others variable. Incl	Included above	already on-site	Included above	Included above Included above	alteady on-site	Included above	half day inspection every 3rd day half day inspection every 3rd day	1 inspection	······································	1 inspection 1 inspection	already on-site	elisento Ol-Line	already on-site	1 inspection	Instantion									R overlap with inspections. Final report only.					TBC		TBC						
																										36																								к					۲								
																				1 8	8	6 16				8 576							8 8	-						•										24 24					_								
																		1 010	1,010	1,010	1,010	5,280	[105,170		82,440						e 650	1,270	570	620	570				570	570									8,640 2.				1 240	040**								
																			. 01	w -	•	v		(plus vehicle) \$		5 4								v						•										s													

5.1.2 Unit Rates Pricing

• The unit rates provided are exclusive of GST.

• The unit rates are inclusive of all costs, fees and charges howsoever arising or incurred in association with the provision of the personnel to Hydro. Refer Table: Summary of Target Cost for separate vehicle charge

• The rates provided are valid for 2 years from Contract execution.

• The unit rates are for work executed from either the Consultant's office and/or Hydro's site and offices.

• For clarity, services which are determined to be variations in accordance with the Contract shall be payable at the rates below unless an agreed lump sum is established between the parties.

ltem	Discipline / Classification	FY2020/21	FY2021/22	FY2022/23
	Administration	Consultant Office (\$/hr)	Consultant Office (\$/hr)	Consultant Office (\$/hr)
1.2.1	Project Director	220	229	239
1.2.2	CQA Manager	130	136	142
1.2.3	Site CQA Engineer	130	136	142
1.2.4	Engineering Support (in-office / Senior Engineer)	140	146	152
1.2.5	CQA Monitor	110	115	120
1.2.6	Technical Specialist	220	229	239

Notes:

• Rates are based on the 8 hr day, 5 day working week in the Draft Construction program provided in Addendum 1

• Rates do not allow for weekend work or night works

• SMEC propose vehicle costs to be charged separately as outlined in the *Table: Summary of Target Cost*. We have listed this separately to offer full transparency to Hydro as it is a significant project cost that is separate to SMEC's professional services as an I.E./CQA.

• Rate increases, as specified in the above table for each relevant FY, shall be applied from 1 July

Where the remuneration method of the proposed Contract is a Schedule of Rates (or similar), any additional costs, expenses, fees, disbursements or charges expressly excluded from the above rates must be specifically detailed in the table below, otherwise they are deemed to have been included and fully incorporated into the above rates.

Item	Qualification and/or Additional Cost	Cost / Unit (note the + is the margin stated in section 5.1.3 below)
1	Courier of samples to lab	cost+
2	3rd party laboratory costs	cost+ Fee estimate of \$405,241
3	variable 3rd party lab costs (eg Destructive testing of seams)	cost+ note: rate depends on quantity

5.1.3 Third-Party costs margin

This section relates to the third party costs specifically set out in section 5.1.2 above.

In relation to these third party costs, the following shall apply:

- (a) where the remuneration method of the proposed Contract is Lump Sum, the rates in this section shall be used to price variation work (i.e. services which have been determined to be a variation under this Deed); or
- (b) A third-party cost includes only those costs from suppliers/contractors engaged by the Consultant and actually incurred by the Consultant, but does not include any costs from the Consultant or its officers, employees, subcontractors and/or subconsultants for which a rate, estimate or budget (including a Labour Unit Rate, Equipment Unit Rate or Material Unit Rate) is already provided in the sections above.

The third-party costs margin stated below shall be applied to the actual invoiced costs of third-party supplies/works/services multiplied by the margin (i.e. the margin is not a 'gross margin'). The margin amount is deemed to cover all of the Consultants profit and overhead costs.

Consultant's third-party costs margin:	5%
--	----

Execution page

Executed as a deed

Signed, sealed and delivered by the Minister for Planning and Public Spaces (ABN 20 770 707 468), in the presence of:

Signature of witness

SAIGHYON YANG -----

Name of witness in full

R DARCY ST, PARRAMATTA NSW 2150 Address of witness

Executed by **Hydro Aluminium Kurri Kurri Pty Ltd (ABN 55 093 266 221)** in accordance with section 127 of the Corporations Act:

Signature of Director

1ARD BROWN

Name of Director

Signature of the Minister for Planning and Public Spaces or delegate

Name of Minister for Planning and Public Spaces or delegate

Signature of Director/Secretary

STEATES JAMES ROBERTS

Name of Director/Secretary

Annexure A Containment Cell Plan

S:10585026_2 BXW

Brett Whitwood

B Kuh



Annexure B Containment Cell Land Plan

Broth Whelesel Sloss

S:10585026_2 BXW



Annexure C Access Road Concept Plan

Brett Whilewel

Jul



SHORT LINE TABLE		
LINE	BEARING	P.O. DISTANCE
1	278°16′45″	20.12
2	8°16′45″	241.4
3	8°16′45″	241.4
4	278°16'45"	(10)
5	278°16′45″	110.64
6	278°16′45″	267.07
7	278°16′45″	20.12
8	278°16'45"	447
9	278°16′45″	804.67
10	8°16′45″	644.14

Annexure D Access Road Certificate of Compliance

Broth Halling P

(Kill B

S:10585026_2 BXW

Access Road Certificate of Compliance

Issued pursuant to clause 3.4(k)(iii)(A) of Schedule 4 of the Voluntary Planning Agreement

Issued to:	Minister for Planning and Public Spaces		
From:	[<mark>#insert name of engineer</mark>]		
Date of Certificate:	[<mark>#insert date</mark>]		
5 - S		a 120 A	ан

The terms used in this Certificate have the same meaning as in the Voluntary Planning Agreement between the Minister for Planning and Public Spaces (**Minister**) and Hydro Aluminium Kurri Kurri Pty Ltd (**Developer**) dated [#insert date] (**Voluntary Planning Agreement**).

- 1. This certificate is issued in relation to the Access Road constructed by the Developer as required by the Voluntary Planning Agreement.
- 2. We warrant and certify to the Minster that:
 - (a) We hold the required skills, experience and knowledge to issue this Certificate.
 - (b) We have carried out inspections of the Access Road in accordance with our obligations under [#insert the document setting out the engineer's obligations to inspect the access road].
 - (c) The Access Road completed by the Developer has been constructed in accordance with, and complies with the requirements of:
 - (i) the Access Road Plans approved by the Minister under clauses 3.4(c) and (d) of the Voluntary Planning Agreement;
 - (ii) the relevant Approvals for the Access Road;
 - (iii) all relevant requirements of the Roads Authority;
 - (iv) all applicable laws and standards; and
 - (v) good industry practice.
- 3. We acknowledge that the Minister is relying on this Certificate.

Signed for and on behalf of [#insert name of engineer]:

Signature

..... Name

Brettwikelieger

Annexure E Stage 1A Works Certificate of Compliance

Broth Wholewood

S:10585026_2 BXW

Auh B Ass

Stage 1A Works Certificate of Compliance

Issued pursuant to clause 6(b) of Schedule 6 of the Voluntary Planning Agreement

Issued to:	Minister for Planning and Public Spaces
From:	[#insert name of Validation Consultant]
Date of Certificate:	[<mark>#insert date</mark>]

The terms used in this Certificate have the same meaning as in the Voluntary Planning Agreement between the Minister for Planning and Public Spaces (**Minister**) and Hydro Aluminium Kurri Kurri Pty Ltd (**Developer**) dated [#insert date] (**Voluntary Planning Agreement**).

- 1. This certificate is issued in relation to the Stage 1A Works constructed by the Developer as required by the Voluntary Planning Agreement.
- 2. We warrant and certify to the Minster that:
 - (a) We hold the required skills, experience and knowledge to issue this Certificate.
 - (b) We have carried out inspections of the Stage 1A Works in accordance with our obligations under [#insert the document setting out the Validation Consultant's obligations to inspect the Stage 1A Works].
 - (c) The Stage 1A Works completed by the Developer:
 - (i) have been constructed in accordance with, and complies with the requirements of:
 - (A) the Voluntary Planning Agreement;
 - (B) the Remediation Consent;
 - (C) the Scope of Works and so as to achieve the objectives of the RAP;
 - (ii) are Fit for Purpose;
 - (iii) are in compliance with all applicable laws and standards; and
 - (iv) are in accordance with good industry practice.
- 3. We acknowledge that the Minister is relying on this Certificate.
- 4. This Certificate operates as a deed poll in favour of the Minister.

Biett Wallion

Luh

Executed as a deed poll in favour of the Minister for Planning and Public Spaces

(If an individual)

Signed sealed and delivered by [insert name] in the presence of:

Signature of Witness

Signature

Name of Witness

Name of signatory

......

.....

(if a corporation)

Executed by **[##] Pty Ltd ACN [##]** in accordance with section 127 of the Corporations Act:

Signature of Director

Name of Director

Signature of Director/Secretary

.....

Name of Director/Secretary

Brothtwholeop

nh